Green Steel Transition

A comparative analysis of Hydrogen-Direct Reduction Pathway Technological Innovation Systems in Korea and Sweden

Danhak Gu

Supervisors

Professor Lars J. Nilsson

Jonas Algers (Ph.D)

Zhenxi Li (Ph.D)

Thesis for the fulfilment of the Master of Science in Environmental Sciences, Policy & Management (MESPOM) jointly operated by Lund University – University of Manchester -University of the Aegean – Central European University

Lund, Sweden, September 2023





Erasmus Mundus Masters Course in Environmental Sciences, Policy and Management





This thesis is submitted in fulfilment of the Master of Science degree awarded as a result of successful completion of the Erasmus Mundus Masters course in Environmental Sciences, Policy and Management (MESPOM) jointly operated by the University of the Aegean (Greece), Central European University (Hungary), Lund University (Sweden) and the University of Manchester (United Kingdom).

© You may use the contents of the IIIEE publications for informational purposes only. You may not copy, lend, hire, transmit or redistribute these materials for commercial purposes or for compensation of any kind without written permission from IIIEE. When using IIIEE material you must include the following copyright notice: 'Copyright © Danhak Gu, IIIEE, Lund University. All rights reserved' in any copy that you make in a clearly visible position. You may not modify the materials without the permission of the author.

Published in 2016 by IIIEE, Lund University, P.O. Box 196, S-221 00 LUND, Sweden, Tel: +46 – 46 222 02 00, Fax: +46 – 46 222 02 10, e-mail: iiie@iiiee.lu.se.

ISSN 1401-9191

Acknowledgements

First and foremost, I am deeply indebted to my supervisors, Dr. Lars J. Nilsson, who provided me with every possible resources for my research journey. And also, to Jonas Algers and Zhenxi Li for their invaluable insights throughout my thesis journey. Thanks to all of you, your warm guidance, this research journey was not only enlightening but also full of excitement leading me to the path of steel industry decarbonization. I also express my sincere gratitude to all interviewees whose names & organizations appear in the paper for their invaluable input.

To my beautiful companions, Borea and Philomene who shared this journey through ups and downs, light and darkness, laughter and tears. And to Jemma, who sprinkled magic on my thesis journey and made it so special until the end. It was truly a blessing having you girls sharing the moments with.

Finally, the most precious, to my family. This final outcome of my 2-year master's program would not have been possible without the tremendous support and unconditional love of my family – my mom, my dad, and my brother. You are my family, my friends, as well as my lifelong mentors. Your love, your presence, your wisdom and your faith in me have been a constant lighthouse and my forever shelter.

Abstract

The steel industry is the world's largest coal consumer and CO2 emitter in the industry sector. When indirect emissions are included, the steel industry is the largest emitter among all other sectors, mainly due to the iron reduction process. However, there are increasing efforts to decarbonize steelmaking technology, e.g., Hydrogen Direct Reduction (H-DR). For instance, Hydrogen Breakthrough Ironmaking Technology (HYBRIT) project in Sweden, - a joint project with SSAB, LKAB, and Vattenfall - succeeded in piloting H-DR and delivered the world's first fossil-free steel. This achievement has inspired other countries to decarbonize their steel industry through a variety of sustainable approaches. The Korean steel industry, particularly POSCO, announced that the company also endeavors to achieve decarbonization through H-DR with its unique technology applied, called HYREX. The H-DR technology adopted in POSCO and in HYBRIT is slightly different, but both companies share the principle of using hydrogen produced from fossil-free energy sources, as an iron reducing agent instead of coal. Nevertheless, getting the technology up and running in Korea is another hurdle, as well as scaling up H-DR in both countries. Therefore, this research aims to examine and compare how the H-DR technology can be diffused in Korea and Sweden to meet the climate target considering the technological innovation system and how they interact to shape the technology development. By analyzing functions that influence the technology diffusion in Korea and Sweden, this research delineates insights into how each country approaches the adoption and diffusion of the technology to meet climate goals based on their different contextual factors. Some highlights from the findings show that Sweden has strong domestic (or EU) cooperation while Korea is much stronger in terms of international cooperation with regards to knowledge sharing network. Moreover, policies, such as Carbon Border Adjustment Mechanism (CBAM) brought all actors in the steel industry, strengthening particularly the public-private partnership in Korea that has been previously lacking compared to Sweden. Domestic market for green steel is stronger in Sweden. Some lessons are drawn from each country: 1) strengthen alliances; 2) enhance international cooperation; 3) revisit policy instruments to support demand-side measures for green steel; 4) maintain consistent investment in R&D; and 5) conduct transdisciplinary research.

Keywords: green steel, industry decarbonization, H-DR, POSCO, HYBRIT, Sweden, Korea, CBAM

Executive Summary

Background & Problem definitions

According to the Intergovernmental Panel on Climate Change sixth assessment report, the largest GHG emissions come from the energy sector (34%), followed by industry sector (24%)¹ in 2019 (IPCC, 2022). Yet, when indirect emissions from electricity and heat production are included, industry sector becomes the single highest GHGs emitting sector imposing energy-intensive industries more burden to decarbonization efforts (ibid). Among the heavy industry sectors, the steel industry uses the largest amount of coal as a fuel, accounting for 8% of global CO2 emissions (IEA, 2020a). Nevertheless, there is a growing effort to decarbonize the steel industry. For example, two steel producers, POSCO – a Korean steel conglomerate, and SSAB – a Swedish steel producer, co-hosted the Hydrogen Iron & Steel Making Forum (HyIS) 2022 in Stockholm, Sweden. At the forum, actors who are involved in steel industry gathered and discussed near zero emission steel transition (HYIS, 2022). The purpose of the forum is to highlight the role of steel industry in reducing GHGs to meet climate targets, focusing on hydrogen iron and steelmaking technology (ibid).

POSCO is a steel producing conglomerate in Korea and the 6th largest global steel producer in 2021 (World Steel Association, 2021). POSCO announced its long-term planning of development of hydrogen steelmaking already in 2010 as a part of its Green Growth Strategy (POSCO, 2010). Since 2016, POSCO has realized the importance of hydroge-reduced steelmaking technology and started mentioning about H-DR process in its annual report (POSCO, 2017a). Soon after, POSCO come up with its own hydrogen iron & steel making technology called HYREX (POSCO, 2022c). In 2020, POSCO made a pledge to achieve net zero by 2050, and in 2022, POSCO announced its plan to enter the demonstration stage of HYREX in 2025 (ibid). The HYREX technology is yet in a theoretical stage, with land acquisition underway for the plant construction (Kyongbuk Ilbo, 2023).

SSAB is the 50th largest global steel producer, located in Sweden (World Steel Association, 2021). The scale of steel production is smaller compared to POSCO, however, SSAB is a pioneer of hydrogen-based ironmaking technology. In 2016, SSAB together with two other companies, Vattenfall – a state-owned electricity utility, and LKAB – state-owned iron ore mining, announced their climate ambition by collaborating the Hydrogen Breakthrough Ironmaking Technology (HYBRIT) project. The project succeeded in operating the pilot plant for sponge iron production in Luleå, Sweden in 2020, delivering the world first fossil-free steel ((Swedish Energy Agency, 2021a). The project plans to enter a demonstration plant to produce fossil free sponge iron on an industrial scale to develop an entire fossil-free value chain for primary steel production (Vogl et al., 2018). Notably, HYBRIT has identified the demand side by teaming up with Volvo Cars to contribute to the creation of a green steel market.

Despite the remarkable low-emission transition technology, there is considerable uncertainty about the technology scale-up and its impact on society. The EU's recent announcement on the Carbon Border Adjustment Mechanism (CBAM) could add a layer of uncertainty for both European and Non-European steel producers and ultimately its effectiveness in achieving carbon neutrality. Given this dynamic context in which both countries are in, it is important to note that all countries are in different economic, geopolitical, social, environmental, and technical status, therefore, there is no absolute one-size fit all solution to achieve the industry's

¹ Global GHG emissions by sector: energy system (34%), Industry (24%), AFOLU (22%), Transport (15%) and Buildings (6%) (IPCC, 2023).

climate goal. All countries have different resource availability, different backgrounds and context. Therefore, this paper aims to identify drivers and obstacles in terms of H-DR technology diffusion in Korea and Sweden.

Research aims and questions

This research paper first seeks to identify the current state of the steel industry in Korea and Sweden in terms of green H-DR and what drives and hinders the technology diffusion and delineate pathways for the further development of hydrogen based green steel to ultimately reach carbon neutrality in both countries. Therefore, the main aim of this paper is to understand and identify the factors through a comparative study between Korea and Sweden which influence the diffusion of green H-DR technology, considering the socio-economic and policy perspectives of technological innovation, and how factors interact to shape and influence the technology development in each country. To better understand the different contexts of Korea and Sweden, research questions are formulated based on the Technological Innovation System (TIS) framework as it helps to identify an internal and external source of dynamics in the structure and functions in a technological innovation system (Bergek, Jacobsson, et al., 2008). The emergence and evolution of the TIS reflects the diffusion of a particular new technology. Using the TIS approach allows to identify the means of diffusion of a technology that has been determined or influenced by entrepreneurs, policy makers and/or other relevant stakeholders (ibid). Among 7 functions, this thesis chose 3 and formulated research questions accordingly (Figure 1-2).

Research design

This research paper chose a qualitative research method to investigate the diffusion of H-DR technology in Korea and Sweden. The choice of this method is motivated by the exploratory nature of this study in order to gain a comprehensive understanding of the drivers and challenges identified by a various set of actors in the steel industry. To ensure the validity, of the findings, this research employed a data triangulation incorporating various sources of data including literature review, interviews with experts in the field, discussions held within steel-related inter/national forum. The decision to use a comparative method along the diffusion of H-DR technology stemmed from the author's motivation to understand diverse backgrounds, contexts and interests in the steel industry that cannot be captured by a technological perspective alone. By conducting a comparative analysis between different regions, Korea and Sweden, this research aims to provide context-specific insights and bring out tailored recommendations for each country which has its own unique opportunities and challenges within its own context.

Results & Implications

The results showed some pronounced differences between Korea and Sweden. With regards to knowledge development whose indicators were the number of R&D and the investment for R&D etc., Sweden is ahead in terms of both the number and the amount of total investment received from Swedish government. Moreover, the domestic collaborations with academia and research institutes have been well settled. Korea, on the other hand, according to the national project platform, there were a couple of joint projects for Green H-DR and the amount of investment from the government was smaller. However, POSCO has been actively engaged in joint international study on H-DR with Australia. Also has organized an international knowledge sharing platform.

In terms of Influence on the direction of search such as additional investment after R&D and policy, the HYBRIT project received additional investment from the EU Innovation Fund with

HYBRIT's shareholders' own cashflow. In Korea, it was not until the EU's announcement on Carbon Border Adjustment Mechanism (CBAM), the cooperation between public-private became more pronounced. However, the amount of investment that the government announced is still smaller considering that the amount of steel production is more than 10 times that of Sweden. In terms of policy, the Industrial Emission Directive, Emission Trading Scheme (ETS) and Eco Design Directive were mentioned by the Sweden/EU respondents, while the Carbon Border Adjustment (CBAM) was mentioned from Korean respondents.

Lastly, in relation to market formation, neither government significantly intervene to create a market for green steel. Instead, it is the steel producers themselves, HYBRIT (or SSAB) and POSCO who have taken the initiative with buyers to establish a value chain within the market. However, the EU acknowledges the important role of the demand side, coming up with the proposal of the revised Ecodesign Directive which tries to include steel. To further boost market formation, governments can play a crucial role as big buyers such as Green Public Procurement. Thus, further discussion and investigation are needed in this regard.

Conclusion & recommendations

Sweden identified key actors and created a strong domestic network to collaborate on R&D with universities, research institutes, etc. Sweden shows that a successful TIS needs stable institutional conditions (Swedish Energy Agency & EU Commission), where actors are supported to gain knowledge and skills from technology experimentation, strong networks (universities, research institutes, corporates, etc.), and invest consistently in learning and policy making process. POSCO, on the other hand, is very active in international collaboration. POSCO has strong international supply chain network with Australia, Southeast Asia, Oman, etc. POSCO endeavers to create a knowledge network platform with other international steel producers to facilitate H-DR whereas a domestic knowledge network was hard to identify. With regards to policies, there are already well formulated policy for the steel industry in Europe, such as Industry Emissions Directive, EU ETS, Ecodesign Directive, but need a room to improve, whereas in Korea, CBAM was the main driver as the EU (6.6%) is one of important POSCO's export market. The market for green steel is not evident, however, a lot of strong signals for green steel demand is emerging which is relatively more pronounced in Swedish domestic market than that in Korea. However, policy tools to support the green steel market formation or green steel price subsidies such as Green Public Procurement, etc., have not been actively considered by either Korea or Sweden. Some recommendations are: 1) strengthening regional alliances; 2) enhancing international collaboration; 3) revisiting policy instruments to support demand-side measures for green steel; 4) maintaining consistent investment and other forms of support in R&D; and 5) conduct cross-disciplinary research.

Table of Contents

L	IST OF FI	GURES	. II
L	IST OF TA	ABLES	. II
A	BBREVIA	TIONS (IF REQUIRED)	III
1	INTRO	DDUCTION	1
	 1.1 PRO 1.2 AIM 1.3 SCO 1.4 ETH 1.5 AUE 1.6 DISE 	BLEM DEFINITION AND RESEARCH QUESTIONS PE AND DELIMITATIONS ICAL CONSIDERATION DIENCE POSITION	3 5 6 7 7
2	LITER	ATURE REVIEW	9
	 2.1 CUR 2.1.1 2.1.2 2.1.3 2.2 THE 2.2.1 	RENT KNOWLEDGE RELATED TO H-DR Iron and steel production Green steel transition and H-DR Policy instruments surrounding green steel: Carbon Contract for Difference, Emissions Trading System, and other demand-side policies CORETICAL & ANALYTICAL FRAMEWORK Technological Innovation Systems (TIS) framework	9 <i>9</i> <i>10</i> <i>13</i> <i>17</i> <i>17</i>
3	RESEA	RCH DESIGN, MATERIALS AND METHODS	. 19
4	 3.1 Res 3.2 DAT 3.3 MET 	EARCH DESIGN 'A COLLECTION METHODS AND MATERIALS COLLECTED 'HODS USED TO PROCESS DATA DUCING COUNTRY CONTEXT	.19 19 21
	4.1 ELE	CTRICITY SUPPLY	22
	4.1.1 4.1.2 4.2 KOR 4.3 TIS 4.3.1 4.3.2 4.3.3	Electricity market structure Hydrogen REAN & SWEDISH STEEL INDUSTRY COMPONENTS Actors Network Institutions	.23 .25 27 30 30 31 32
5	FINDI	NGS	.34
	5.1 FUN 5.2 FUN 5.2.1 5.2.2 5.2.3 5.2.4 5.3 MAH	ICTION 1 KNOWLEDGE DEVELOPMENT & DIFFUSION ICTION 2 INFLUENCE ON THE DIRECTION OF SEARCH Investment Industrial Emissions Directive (IED) Emissions Trading System (ETS) Carbon Border Adjustment Mechanism (CBAM) RKET FORMATION	34 37 38 38 39 41 41
6	DISCU	SSION	.46
	6.1 Сом <i>6.1.1</i>	IPARISON OF THE KOREAN AND SWEDISH H-DR INNOVATION SYSTEMS Discussion to RQ1: How has knowledge development and diffusion been supported in Korea VS Swede	46 m? 46
	6.1.2	Discussion to RQ2: How has the guidance of search been supported in Korea and Sweden?	47

6.1.3	6.1.3 Discussion to RQ3: How has the market formed for the green steel in Korea and Sweden?					
6.2 Rei	FLECTION ON MY STUDY	49				
6.2.1	Methodological and analytical choices	49				
6.2.2	Legitimacy					
7 CONC	CLUSIONS	51				
7.1 Rec	COMMENDATIONS FOR FUTURE RESEARCH	52				
BIBLIOGR	АРНҮ	54				
APPENDIX	APPENDIX					

List of Figures

Figure 1-1 Final Energy demand of the heavy industry sector by fuel (unit: N 1	Atoe per year)
Figure 1-2 Selected TIS for the study	6
Figure 2-1 Primary production (BF-BOF) & H-DR, and secondary produc 9	tion pathways
Figure 2-2 DRI and some stacks of experimental HBI behind	12
Figure 2-3 Basic functioning of CCfD	14
Figure 4-1 Korea's electricity market structure	23
Figure 4-2 Sweden electricity market structure	24
Figure 4-3 POSCO integrated steel mills locations	27
Figure 4-4 BF and FINEX process comparison	
Figure 4-5 HYBRIT's H-DR facilities locations	29
Figure 7-1 Comparison H-DR TIS between Sweden & Korea	

List of Tables

Table 1-1 TIS functions and indicators	4
Table 2-1 EU-ETS & K-ETS	15
Table 3-1 Interviewees for this research	20
Table 4-1 Green hydrogen production target comparison	26
Table 4-2 Hydrogen price target in Korea and Sweden	26
Table 4-3 POSCO plants in Pohang and Gwangyang	
Table 4-4 SSAB's BF plants	
Table 5-1 HYBRIT work packages and partners	35
Table 5-2 Korean national project for H-DR from NKIS	
Table 5-3 Number of research and the investment	

Table 5-4 Timeframe of EU ETS & K-ETS	41
Table 5-5 Comparison between Korea & Sweden	44

Abbreviations (if required)

H-DR	Hydrogen-Direct Reduction
HYBRIT	Hydrogen Break Through Iron making Technology
HYREX	Hydrogen Reduction Ironmaking
TIS	Technological Innovation System
CBAM	Carbon Border Adjustment Mechanism

1 Introduction

Throughout history, iron and steel have had a significant impact on people's lives. Iron and steel have been indispensable, playing a crucial role in shaping human civilization, the rise and fall of global power dynamics between countries, for example through the production of weapons, agricultural tools and so on. In modern society, the role has become even more important in everyday life, including transport, construction, household appliances, and many more. Despite its importance to human society, the steel industry is the largest coal using industry as it is a reducing agent during the ironmaking process (Figure 1-1) and overall is responsible for about 7-8% of global carbon dioxide emissions (IEA, 2020a). The latest report of the Intergovernmental Panel on Climate Change (IPCC) highlights that the primary sources of GHG emissions are energy systems accounting for 34%, followed by the industry sector at 24% (IPCC, 2022).² Nevertheless, when considering both the direct emissions from steel industry itself and its indirect emissions such as electricity and heat production that attributed to the industry sector, the GHG emissions share of the industry sector becomes equal that of the energy supply sector (34%) making abatement efforts to decarbonize energy-intensive industries more complicated (ibid).



Figure 1-1 Final Energy demand of the heavy industry sector by fuel (unit: Mtoe per year)

Source: IEA, Final energy demand of selected heavy industry sectors by fuel, 2019, IEA, Paris https://www.iea.org/data-and-statistics/charts/final-energy-demand-of-selected-heavy-industry-sectors-by-fuel-2019, IEA. Licence: CC BY 4.0

The steel industry is the backbone industry in both Korea and Sweden; however, it is one of the largest GHG emitters in both countries, being responsible for 11% of the Sweden's total emissions and 15% of the Korea's total emissions (Jernkontoret, 2018; Korea Institute for Industrial Economics & Trade, 2022). This highlights the pressing necessity to decarbonize the steel sector by identifying opportunities and challenges for green transition of the steel industry, as well as addressing gaps between theory and practice. In order to achieve decarbonization, it

² Global GHG emissions by sector (IPCC, 2022): Energy system (34%), Industry (24%), AFOLU (22%), Transport (15%) and Buildings (6%).

is necessary to consider countries' own characteristics and develop context-specific plans and strategies.

There is a growing number of efforts to reach decarbonization in the steel industry both domestically and globally. For example, in 2022, two global steel producers, POSCO – a Korean steel conglomerate, and SSAB – a Swedish steel producer, co-hosted the 2^{nd} Hydrogen Iron & Steel Making (HYIS) Forum 2022 in Stockholm, Sweden (HYIS, 2022). The forum brought together relevant actors - including steel producers, international organizations, hydrogen producers, steel consumers, engineers, academia, researchers, etc. – to discuss the transition to hydrogen-based near zero emission iron and steelmaking technology to highlight the role of the steel industry to meet the climate targets (ibid).

POSCO started as a state-owned steel producer and later privatized in 2000 (POSCO, 2015). It then expanded its business to battery materials, lithium/nickel, hydrogen, energy, construction, and agri-bio, and became POSCO Holdings in 2022 (POSCO Holdings, n.d.). The company is the 6th largest global steel producer in 2021 accounting for 42.96 Mt (World Steel Association, 2021) and announced its 2050 carbon neutrality roadmap in December, 2020 (POSCO, 2020). The roadmap sets out the medium and long-term targets: in short-, mid-term, POSCO is focusing on productivity and energy efficiency by optimizing the raw material mix and the streamlining of operations such as increase the use of scrap in the Blast Furnace - Basic Oxygen Furnace (BF-BOF) process and using H2-rich off-gas as a reducing agent in the existing blast furnace (BF) as well as scaling up Carbon Capture Utilization and Storage (CCUS) technology. The long-term plan is to develop the hydrogen-based ironmaking technology called HYREX and to commercialize the hydrogen-based steelmaking process (ibid). In 2022, POSCO announced its plan to enter the demonstration stage of HYREX from 2025 (POSCO, 2022a). POSCO's HYREX technology is still at a theoretical stage, with land acquisition for the plant construction underway (Kyongbuk Ilbo, 2023).

SSAB is the 50th largest global steel producer, located in Sweden ((World Steel Association, 2021). The annual crude steel production is 8.18 Mt in 2021 (ibid), but SSAB, together with its partner companies, is a pioneer in the hydrogen-direct reduction ironmaking technology. In 2016, after the Paris Agreement, SSAB together with two other Swedish companies, Vattenfall - a state-owned electricity utility, and LKAB – state-owned iron ore mining, announced their climate ambition by collaborating the Hydrogen Breakthrough Ironmaking Technology (HYBRIT) project (Pei et al., 2020). In 2018, the project constructed a pilot plant at the SSAB site in Luleå, Sweden and it succeeded in operating the pilot plant for sponge iron production in 2020, becoming the world's first fossil-free steel producer (ibid). The project aims to enter the demonstration stage by 2025 to produce fossil free sponge iron on an industrial scale to develop an entire fossil-free value chain for primary steel production (Åhman et al., 2018). Moreover, Volvo Cars joined the project and collaborated with the HYBRIT consortium, becoming the first customer of fossil free steel (SSAB, 2021).

Although Sweden's success in piloting the H-DR process has clearly shown the world that the steel industry is not a "hard-to-abate", but can be a "fast-to-abate" sector (Agora Industry & Wuppertal Institute, 2023), the actual application and diffusion of the technology may be a different story for other steel producers with different contexts and environment. Therefore, it is necessary to understand the unique characteristics of each country, to examine the factors that lead some countries to successful technology development while understanding the disparities in other countries, to learn lessons from each other, and ultimately to develop context-specific decarbonization plans and strategies.

1.1 Problem Definition

Despite the ambitious plan and collaborative efforts of the steel producers to decarbonize their steel making process, there is a considerable uncertainty about how the H-DR technology can be scaled up and contribute to carbon neutrality while maintaining competitiveness in the market (Vogl et al., 2021). For example, uncertainties remain in areas such as securing renewable electricity supply, grid connectivity, the electricity cost associated with electrolysis to produce green hydrogen for H-DR process, storage of hydrogen, hydrogen cost, investment, trade conflicts and more (Åhman et al., 2018; Kang, 2022; Korea Institute for Industrial Economics & Trade, 2022; Kushnir et al., 2019). In the broader global landscape, the potential impact on climate and international trade of the EU's introduction of the Carbon Border Adjustment Mechanism (CBAM) and the US Inflation Reduction Act (IRA) is yet to be fully anticipated, making it difficult for the steel industry to navigate and adopt adaptive strategies (KEEI, 2022; Project Syndicate, 2022). Given this dynamic context, it is important to acknowledge that each country possesses its own unique economic, geopolitical, social, environmental, and technical conditions. This demonstrates that technology diffusion is shaped by the dynamic influences of various factors, highlighting the importance of considering different dimensions of the sociotechnical landscape, encompassing social, economic, and political views.

To be more country- and company-specific, POSCO has published a sustainability report since 1994, with the topic of the hydrogen steelmaking first mentioned in the 2008 report (POSCO, 2008). It was not until 2016 that the company began to address this matter with greater emphasis (POSCO, 2008, 2021a). Despite its early interest in hydrogen-based steelmaking, its HYREX concept was not pronounced publicly until CBAM announcement around 2021. Instead, POSCO was known for being the largest GHG emitter among Korean corporates for the past 10 years, with POSCO alone accounting for 11.5% of total national GHG emissions in 2021 and is still expected to receive the free allowances until 2025 (GIR, 2022; Greenpost Korea, 2022). Although POSCO announced its plan to achieve carbon neutrality by 2050 and its plan for HYREX, details need further clarification, and progress has been relatively slow.

For the HYBRIT initiative, despite Sweden's relative abundance and access to renewable electricity, some concerns are mentioned, such as securing a large amount of renewable electricity for electrolysis for the H-DR process, fluctuating electricity prices, ensuring sufficient transmission capacity of renewable electricity demanded by the H-DR and by other green industrialization initiatives, and grid connection as the technologies scale up (Åhman et al., 2018; Kushnir et al., 2019; Swedish Energy Agency, 2021a). In addition, there have been some worrying voices about excessive public investment in green steel as well as about the potential neglect of the rights of the indigenous Sami people (Henrekson et al., 2021; McVeigh, 2022).

Considering the different pace of H-DR technology diffusion as well as the distinct challenges faced in Korea and Sweden, a comprehensive understanding of each country's own features is prerequisite to formulate plans and strategies that harmonize with their distinct contexts. Therefore, given this significant influence of the steel industry in both countries and their efforts to develop the H-DR technology to meet the climate target, this research focuses on comparing the H-DR development in the Korean and Swedish steel industry, more specifically HYBRIT and POSCO, with an innovation system approach.

1.2 Aim and research questions

The aim of this research is to complement the growing body of knowledge on the H-DR development by applying an innovation systems approach in Korea and Sweden and to contribute to the existing literature on the steel industry decarbonization. By examining the

dynamics of technology development and diffusion in these two different geographical locations, this study seeks to understand their different contexts and different approaches to the technology development and diffusion. This research aims to identify the internal and external environments of Korea and Sweden that contribute to accelerating and hindering the diffusion of H-DR as well as the opportunities and challenges that Korea and Sweden face in their green steel transition.

To better understand the different contexts of Korea and Sweden, research questions are formulated based on the Technological Innovation System (TIS) framework as it helps to identify internal and external sources of dynamics in the structure and functions of a technological innovation system (Bergek, Hekkert, et al., 2008). The emergence and evolution of the TIS reflect the diffusion of a particular new technology (ibid). Using the TIS approach, it is possible to identify means of accelerating the diffusion of a technology that have been determined or influenced by entrepreneurs, policymakers and/or other relevant stakeholders (ibid). Bergek, Hekkert et al. (2008) have outlined seven functions and indicators as Table 1-1:

System function	Examples of indicators / Data types		
1. Knowledge development and diffusion	R&D projects, Patents, Investments in R&D, Workshops and conferences, Size, and intensity of learning networks		
2. Influence on the direction	Factor / product prices (e.g., taxes and prices in the energy)		
of search	Regulatory pressures (e.g., quota systems for renewable electricity)		
	Government / industry targets regarding the use of a specific technology		
	Estimates of future growth potential		
	Articulation of interest by leading customers		
3. Entrepreneurial	Number of new entrants.		
experimentation	Number of experiments with the new technology.		
	Degree of variety in experiments (e.g., number of different applications)		
4. Market formation	Number, size, and type of markets formed.		
	Timing of market formation		
	Drivers of market formation (e.g., support scheme)		
5. Resource mobilization	Volume of capital and venture capital		
	Volume and quality of human resources (educational data)		
	Volume and quality of complementary assets		
6. Legitimation	Attitudes towards the technology among different stakeholders		
	Growth of interest groups		
	Lobbying activities		
	Political debate		
7. Development of positive	Strength of political power of TIS actors.		
externalities or "free utilities"	Activities aiming at uncertainty resolution.		
	Existence / development Pooled labor market		

Table 1-1 TIS functions and indicators

		Information and Knowledge flows
--	--	---------------------------------

Source: Bergek et al., 2008

All indicators and their functionalities can be crucial to examine, however, this study accentuates more on "knowledge development", "influence on direction of search," and "market formation" than the other functions; consequently, research questions derived directly from these functions were formulated. The research questions are as follow:

RQ 1.1 How has knowledge development and diffusion been supported in Korea VS Sweden?; RQ 1.2 How has the influence on the direction of search been supported in Korea VS Sweden?; RQ 1.3 How has the market formed for the green steel in Korea VS Sweden?

1.3 Scope and Delimitations

The research focuses on the decarbonization of the steel industry in Korea and Sweden, where the steel industry plays a central role in economies of both countries. There are several different pathways to decarbonized steel production, including hydrogen-based steelmaking, Carbon Capture Usage and Storage (CCUS), scrap, steel substitution, etc. (Bulkeley et al., 2022). This paper concentrates on the hydrogen-based iron and steel making, also known as Hydrogen Direct Reduction (H-DR), as the technology is seen as a promising enabler to realize the respective countries' fossil-free plans by 2030 in Sweden and by 2050 in Korea (POSCO, 2021a; SSAB, n.d.). The focus on Korea and Sweden was chosen for several reasons: 1) the steel industry is an important backbone industry contributing to the economies of both countries; 2) the steel industry is the largest GHG emitter in both countries, hence the urgency of decarbonization; 3) both Korea and Sweden are committed to developing H-DR; and 4) Sweden, specifically the HYBRIT initiative, is a pioneer in H-DR. This paper examines the decarbonization plans of the steel industry in Korea and Sweden, respectively, with a focus on their individual approaches rather than their interrelationships. Although there are various issues from the global landscape, i.e., overcapacity, scrap availability, grid connectivity, etc., the study remains dedicated to an assessment of the H-DR innovation system as a first step in my research on the decarbonization of the steel industry.

The theoretical and analytical approach of the thesis uses is based on the Technological Innovation System (TIS) framework, which was devised by Hekkert et al. (2007) and soon updated by Hekkert & Bergek & Jacobsson (2008). When it comes to investigating innovation systems, it often uses statistical approaches and overlooks the qualitative aspects of process and information (Hekkert et al., 2007). To solve this, Hekkert et al. proposed the TIS framework that highlights the most important processes that need to take place in innovation systems to succeed in technology development and diffusion. The purpose of using functions of the TIS concept is to understand processes of technological change and innovation (Hekkert et al., 2007; Bergek, Hekkert, et al., 2008; Bergek, Jacobsson, et al., 2008). The choice of applying the TIS framework in this research was motivated by the need to analyze technological change and innovation comprehensively, taking qualitative factors that are often neglected, into account to provide holistic picture of the hydrogen-based steelmaking technology and its surrounding environment (Hekkert et al., 2007). Though the TIS framework is a complete package with seven functions and their indicators, due to the time constraints and some confidentiality that surrounds the technology, not all seven functions could be answered. Instead, the author had to select and tailor some of the functions to focus more on understanding the internal and external environment and its influence on technology development as Korea and Sweden have different socio-economic context and climate strategy around the steel industry and its H-DR technology. Thus, as a first step in analyzing the steel industry decarbonization, the author

needed to thoroughly understand the external surroundings of the H-DR holds in two countries. I focused on the "direction of search" (in particular, policy) and "market formation" functions to gain insights into how the external factors have shaped the technology diffusion initiative and the process. Secondly, the "knowledge development" function was chosen to understand the internal factors of two countries. R&Ds in knowledge development is essential in technological innovation as it is the foundation of technological innovation and then used in the production of final goods and leads to increases in growth and enabler of sustainable economic growth (Ulku, 2004). Therefore, due to the aforementioned reasons, functions that are closely related to Function 1. knowledge diffusion, Function 2. guidance on research and Function 4. market formations were prioritized to examine in this paper.



Figure 1-2 Selected TIS for the study

Source: Author's own based on Bergek et al. (2008)

To enhance the reliability as well as validity of the thesis, triangulation of data collection was used i.e., data from grey and white documents, interview, and observation. One of important thesis aims was to capture various stakeholder perspectives who are either indirectly or directly involved in steel space to understand the overall steel industry decarbonization process. Eight interviews were conducted, four from Korean side and four from Swedish side. This process approach conceptualizes development and change processes as sequences of events that occur within the technology specific innovation system under investigation. These events may include seminars on technology, the startup of R&D projects, announcements of the availability of resources and so forth.

1.4 Ethical Consideration

There were no conflicts of interest, and this research was conducted independently by the author on the topic of H-DR technology innovation systems in Korea and Sweden. No one was in the position to influence my analysis and conclusions other than my supervisors. Contributions towards field work expenses in Luleå, Gällivare, and Boden in Sweden were provided by the Central European University in Vienna, Austria, and Lund University. There was no outside organization or third party funding this research. Participation in the interviews was voluntary, and the participants were offered the choice of anonymity before the interview began and were given sufficient time to consider their participation. Participants were kept anonymous upon request. Data will be carefully stored under the Lund University system. As a researcher, I will ensure non-disclosure, and no harm and comply with all necessary regulations, such as the General Data Protection Regulation (GDPR). Interviewees had the opportunity to screen their responses on the paper prior to the publication upon request. This will include an explanation of the research project, its aim and assurance of confidentiality, and the right to withdraw from the interview. If there is any confidential or sensitive data shared, it will be kept completely undamaged and stored either on the hard drive of a password-protected or in the LU Box.

1.5 Audience

This thesis is intended for a wide audience, including experts, academics, and policymakers interested in industrial decarbonization, especially the steel industry. Given the context-specific character of this comparative study, the actors who are likely to find this thesis useful include stakeholders in Korea and Sweden who are working in the steel industry as well as members of civil society and non-governmental organizations who are working on steel industry decarbonization. It is hoped that this research would offer specialists in the fields of energy and industry, as well as policymakers, with insight that will help achieve carbon neutrality by 2050.

1.6 Disposition

Chapter 1 illustrates the topic of H-DR and the general background of H-DR technology adopted by Korea and Sweden and its importance for the decarbonization of the steel industry in both countries. It then presents the current status and the distinct challenges in H-DR transition in both countries. It further introduces the general characteristics of this research including the research aim, research questions, delimitations and ethical considerations and audience, etc.

Subsequently, Chapter 2 provides a more in-depth analysis of the literature on the green steel transition focusing on the H-DR process as well as conventional steelmaking, i.e. the blast furnace route. It also explains the theoretical and analytical framework that this uses for its analysis and its justification.

Chapter 3 describes a detailed description of the research design, introducing the advantages of using qualitative research method and justification of its use for this study. The methods for data and material collection are introduced, followed by a detailed description of the interview respondents.

In addition to the literature review on existing literature, Chapter 4 further esplains a comprehensive context of each country in terms of its electricity system and its steel industry as a groundwork to better understand aspects that influenced their respective initiatives, positions, etc.

Chapter 5 of this study presents the findings of the interviews, search and observation, providing insight into different perspectives and elements of the H-DR process in each country. The findings are examined in accordance with the tailored TIS framework.

Chapter 6 presents the discussion in which the findings from the interview, search and observation are further discussed and interpreted and placed in the context of existing literature and research. After, the author articulates reflections of this research.

Finally, in the last chapter, conclusions and recommendations are proposed summarizing the overall research, highlighting the importance of the research, and also suggesting further research in this area.

2 Literature Review

This chapter illustrates the literature review, followed by the theoretical & analytical frameworks that the research is based on.

2.1 Current knowledge related to H-DR

2.1.1 Iron and steel production

Simply put, steel is an alloy of iron and carbon that is the most widely used material in the world (Bulkeley et al., 2022; Vogl, 2023). Steel can be produced from two different raw materials: iron ore and steel scrap. Primary production is the process of manufacturing steel from iron ore, while secondary steel production is the recycling of scrap (Vogl, 2023). The distinction between primary and secondary production, however, can become less definitive, since scrap is also used as a feedstock in primary production and virgin iron can be used in secondary production, with the blast-furnace - basic oxygen furnace (BF-BOF) route being the most common method, accounting for around 70% of global steel production and 90% of primary production (IEA, 2020a). Figure 1-3 shows the three main stages of steel production: raw material preparation, ironmaking, and steelmaking (ibid). Following mining and beneficiation, iron ore requires further processing prior to the ironmaking stage. Iron ore fines need to be agglomerated through sinter or pellets (ibid). The ironmaking stage involves two chemical reactions: the reduction reaction and the melting reaction.

Coal serves as the main energy source for the steel industry, accounting for around 75% of its energy demand (IEA, 2020) and is referred to as metallurgical coal (Vogl, 2023). Some of its main roles are as: a fuel to supply heat to the plant, and a chemical reducing agent to convert iron ore (iron oxide) into iron, etc. (D1 ez et al., 2002; Vogl, 2023). The steel industry is currently the biggest industrial consumer of coal followed by the power sector (Vogl, 2023).



Figure 2-1 Primary production (BF-BOF) & H-DR, and secondary production pathways

Source: Author's own based on IEA (2020) and Vogl et al. (2021)

Blast Furnace - Basic Oxygen Furnace (BF-BOF) Route

The blast furnace is the most dominant plant used in ironmaking stage for the primary production (IEA, 2020). The blast furnace is a mature production system that has been used for the last thousand years, and fossil coke has unique desirable mechanical properties making it hard to replace with less emitting alternatives (Vogl et al., 2021). One reason why the steel industry is often labeled as 'hard-to-abate' sector is that the majority of steel industry emissions come from the ironmaking stage, where fossil coke is used to reduce iron in the blast furnace (Vogl et al., 2021). During the reduction reaction, when iron ore (Fe2O3) meets coke (CO), liquid iron (Fe), known as hot metal or pig iron, and carbon dioxide (Co2) are produced, and this is where the highest direct CO2 emissions are generated in the steel industry (Korea Institute for Industrial Economics & Trade, 2022; V. L. Vogl, 2023). The hot metal produced is put into the basic oxygen furnace (BOF), where oxygen is injected to decrease the carbon content of the steel produced. Following that, the liquid steel produced in the BOF is subsequently casted, rolled and processed into steel products (Suopajärvi et al., 2017; IEA, 2020).

Due to the steel industry's dependence on metallurgical coal, primary steel production, especially the BF-BOF route, is by far the largest source of GHG emissions from the steel industry. Blast furnaces have relatively long lifetimes, about 17 years, thus, investment decisions to construct new or refurbish existing plants can lock-in significant emissions and future climate change (Vogl et al., 2021). Vogl, Olsson, and Nykvist emphasized that substantial changes must be made to the primary production process in order to decarbonize the steel sector. These changes can be made either by redesigning the blast furnace with a lower-emission application or by replacing the blast furnace with another technology, such as the H-DR process, which has recently gained significant attention in the steel industry (Vogl et al., 2021). Subsequent to the Paris Agreement, several steel producers have announced their plans for green steel (ibid). However, due to the geographical variations, predominantly in Asian countries with relatively young and large-scale BF-BOF route makes it difficult to rapid transition and maintain industrial competitiveness (Kim & Sohn, 2022).

2.1.2 Green steel transition and H-DR

Until 2010, EU climate policy for energy-intensive industries was mostly focused on efficiency improvements and marginal emission reductions (Åhman & Nilsson, 2015; Vogl et al., 2021). This was done by conserving industrial structure rather than encouraging the systemic change that is necessary to achieve zero emission (ibid). Following 2010, the emphasis shifted to developing technology and innovation to enable deep decarbonization. Innovation funding programs such as Horizon 2020, EU Innovation fund, etc., have fostered technical innovation and the scale-up of novel technologies (Vogl et al., 2021).

As new regulations imposed on emission-intensive industries as well as increasing sustainable investment have contributed to a growing interest in 'green steel', there is a growing body of existing literature on low and zero emission steel production. Green steel is commonly referred to as a strategic approach aimed at making the steelmaking process greener and more sustainable (SSAB, n.d.). According to Griffin & Hammond (2021), green steel refers to a steelmaking process designed to reduce GHG emissions, potentially cut costs, and improve the quality of steel in contrast to conventional processes (Griffin & Hammond, 2021). However, there is no precise definition of green steel yet, rather, it encompasses many different low and zero emission pathways. As the processes of procurement and manufacturing include a wide variety of stakeholders across the globe, there is a different understanding of what 'green steel' really is (Anderson, 2022). Therefore, to understand whether the final steel product falls into the 'green'

category, steelmaking technologies, R&Ds, emissions reduction initiatives, and each step of the production process need to be thoroughly investigated (ibid).

The European Steel Association, EUROFER, has defined two pathways for low CO2 steel production: Smart Carbon Use (SCU) and Carbon Direct Avoidance (CDA) (European Steel Association, 2019; Anderson, 2022). SCU involves modifications to existing iron and steelmaking plants that still rely on fossil fuels that help reduce the use of CO2, such as Carbon Capture and Usage (CCU) and Carbon Capture and Storage (CCS), while CDA includes hydrogen-based metallurgy, generated from renewables, to replace coal as the main reduction agent for ironmaking stage (ibid). However, the EUROFER roadmap did not define what green steel is, nor did it explicitly categorize the extent to which low or zero CO2 pathways, based on different technologies, qualify as green steel. Muslemani et al. (2021) pointed out that the research or consensus on concept of green steel remain lacking. The definition of green steel has been seldom used in the academic literature and when its used, its meaning is broad (Muslemani et al., 2021). The existing literature on green steel indicates a gap in the definition of what green steel is, therefore, it shows the need for further clarity in categorizing pathways that qualify as low or zero emission pathways in primary or secondary production.

In addition to the issue of varying interpretations of green steel, the steel industry encounters the problem of cost competitiveness of lower emissions steel. In spite of the substantial investment allocated to R&Ds on technical innovation, there are uncertainties around the business case for green steel production (Kushnir et al., 2019; Vogl et al., 2021). The reason for the lack of widespread adoption of low-emission steel production is that the lack of certainty on cost competitiveness both in Europe and non-Europe, requiring the provision of public support for its implementation (ibid). This is where active policy can come and play a significant role in providing support to the green steel (Vogl et al., 2021). Vogl et al. (2021) conducted an extensive analysis to identify the policy instruments available for the early commercialization phase of green steel including both the supply and demand sides.

Hydrogen-Direct Reduction (H-DR)

To completely decarbonize primary production, GHG emissions must be entirely avoided or must be captured and stored (Vogl et al., 2021). One of GHG-avoided pathways for primary production that can replace blast furnace is hydrogen direct reduction (H-DR). In this pathway, hydrogen, that is produced through electrolysis from renewable electricity, replaces the use of coke as a reduction agent. Consequently, when iron ore (Fe2O3) meets hydrogen (H), water (H2O) is generated as a by-product and no carbon dioxide is produced (Åhman et al., 2018). The H-DR process typically involves the feeding of pre-heated iron ore into a reduction shaft, where it is converted to direct reduction iron (DRI) and then further compacted to produce hot briquetted iron (HBI). HBI is then fed into an electric arc furnace (EAF) and melted and transformed to liquid steel (Vogl et al., 2018). As the H-DR process is expected to be entirely electrified, the emissions are largely determined by the power grid emission intensity (GEI) (ibid). The decarbonized electricity required for hydrogen production and EAF operation therefore, needs to be ensured (Kim & Sohn, 2022). Although decisions for green steel is pivotal and inevitable, many challenges arise as achieving deep decarbonization requires not only technology innovation but also a systemic change of the steel production process (Vogl et al., 2021). This underscores the industry's commitment to advancing lower-emission technologies, addressing environmental concerns while navigating complex global dynamics. Additionally, although the H-DR process is considered as a fossil-free alternative, and CO2 emissions from the BF-BOF route can be avoided in the H-DR route if renewable electricity is available, it is not completely CO2-free. CO2 emissions are nevertheless still occur in processes

such as the mining and manufacturing of iron ore and limestone, lime calcination and the addition of carbon as an essential component of steel (ibid).

At present, H-DR is still under development, and the H-DR process can vary from company to company. Broadly speaking, it can be divided into two main process methods: the shaft furnace method used by HYBRIT and the fluidized-bed reactor adopted by POSCO (Kang, 2022). There are three main differences: difference in materials, in methods, and in carbon emissions (POSCO, 2022a). A shaft furnace uses pellets, iron ore that has been processed into a round shape of a certain size, while a fluidized bed reduction reactor uses iron ore fines directly, which do not require processing (ibid). In terms of method, shaft furnace use hydrogen which is a high-temperature reduction gas, moves from the bottom to the top through the empty spaces between pellets (ibid). These pellets then become DRI by going into the reduction furnace. In a fluidized bed reactor, high-temperature reduction gas is evenly spread through a perforated grid at the bottom of the reactor to mix iron ore fines and start a reduction process (ibid). After going through several steps of reactors, the reduced iron ore fines become DRI. According to the POSCO article (2022), the production of 1 ton of pellets in the absence of renewable energy source can result in 50-150kg of CO2 emission (POSCO, 2022a). However, fluidized bed reactors are not required to produce pellets, as iron ore is used directly without any processing (ibid). The HYBRIT project has set a good example by successfully piloting H-DR and demonstrating the feasibility of fossil-free production to a global audience. The fluidized bed reactors method that POSCO adopted has the potential to exhibit energy and material efficiency, nonetheless its implementation and operation have not yet been realized. Regardless of where the H-DR process stands today, there are several uncertainties surrounding the actual scale-up of the H-DR process.



Figure 2-2 DRI and some stacks of experimental HBI behind

Source: Author's own, photographed at HYBRIT in May 2023

The HYBRIT project has set a good example in terms of H-DR technology and showed the world that steel industry can go fossil-free. SSAB, LKAB and Vattenfall have collaborated HYBRIT project with funding from the Swedish Energy Agency and EU's Innovative Fund. In early 2016, these three companies announced to implement a fossil-free steel production process in Sweden and a pre-feasibility study was initiated (Åhman et al., 2018). In 2017, a research program was set up with the help of the Swedish Energy Agency. Many different partners such as universities and research institutes have been able to engage in this process, such as Luleå University of Technology, Lund University, KTH, Stockholm Environment Institute (SEI), etc. Therefore, knowledge development for H-DR in Sweden has been supported by many different actors (HYBRIT, n.d.).

For Korea, on the other hand, it was hard to find literature, reports or articles that address funding sources for technology development and its associated research or knowledge sharing in Korea. Only a few domestic cooperations with different actors, networks or institutions inside Korea for H-DR development were identified to a limited extent until 2023, the year when reporting under the Carbon Border Adjustment Mechanism (CBAM) comes into effect. It was only some articles published by POSCO described that POSCO will work with the government and Korean steel companies to promote hydrogen-based steelmaking but did not specify. Overall, it gives a strong impression that POSCO is a lonely wolf that needs to pave its way on its own. However, POSCO has been actively seeking cooperation with international actors outside Korea, particularly Australia. For instance, POSCO Group plans to invest in Australia in renewable energy and water electrolysis for hydrogen manufacturing and to manufacture hot briquetted iron (HBI) (POSCO, 2022b). Moreover, POSCO has initiated the first international Hydrogen Iron & Steel Making Forum (HyIS) and it provided a platform for international steelmakers and stakeholders to gather and share knowledge on decarbonized hydrogen steelmaking technology (HYIS, 2022). Such an initiative has fostered knowledge development on a global scale, although knowledge development among academia, firm and government appears to be lacking inside Korea. Recently, the Korean Ministry of Trade, Industry and Energy (MOTIE) announced the Korean steel industry development strategy for the transition to lowcarbon steelmaking to respond to the urgency of addressing CBAM (MOTIE, 2023b). It involves R&D investment, publishing steel industry roadmap for the carbon neutrality establishing scrap industrial ecosystem (ibid).

2.1.3 Policy instruments surrounding green steel: Carbon Contract for Difference, Emissions Trading System, and other demand-side policies

Carbon Contract for Difference (CCfD)

Vogl et al. (2021) conducted a comprehensive policy evaluation of supply-side production subsidies and of demand-side market creation policies for the early-commercialization phase of the green steel technology (Vogl et al., 2021). They introduced several policies, including CCfD as a part of a supply-side policy. A CCfD is an EU subsidy agreement that compensates for the difference between the strike price, i.e. the agreed price in the contract between the regulator and a steel producer, and the yearly average price of emissions allowances, i.e. EUAs for a decarbonization project (EUROFER, 2021). It is a project-based financial instrument that provides a fixed carbon price usually over a long-term period around 20-30 years for decarbonization projects (ENGIE, n.d.; Vogl et al., 2021). When carbon prices in markets are lower than the strike price, the government compensates for the difference for each ton of

reduced emissions realized by the project in comparison to the conventional technology. When the carbon price is higher than the strike price, the project pays back to the government for the disparity (Figure 2-2) (Richstein & Neuhoff, 2022). This policy applies to the steel industry that allows the steel industry receive the difference between the strike price and the average annual carbon price in the ETS for each ton of CO2 that remained unemitted as a result of opting for a lower-carbon investment option (Vogl et al., 2021). CCfD can be an effective tool by enabling the provision of carbon prices above the level of market prices. Additionally, it can mitigate carbon price risks, hence reducing financing costs. Thus, CCfD can facilitate the realization of the first industrial-scale production of low-emission steel (Richstein & Neuhoff, 2022; Vogl et al., 2021). Nevertheless, there are several concerns that CCfDs alone may not be enough in mitigating the investment risks associated with low-emission primary steel production, particularly for producers moving towards electrification, where electricity prices drive production costs (ibid). A stable carbon price from a CCfD may not always guarantee a business case for green steel, therefore, government support for long-term electricity price arrangement is needed to supplement CCfD (ibid). Most EU energy-intensive industries already get tax and renewable levy exemptions for electricity production (Åhman & Nilsson, 2015; Vogl et al., 2021). In the context of Korea, it has been observed that Korean industries also benefit from low industrial electricity prices (IEA & KEEI, 2022). The industrial electricity price in Korea is among the lowest in OECD countries, with USD 95/MWh in 2019 (ibid). Moreover, additional policies may be required to effectively reduce emissions from energy-intensive materials production, such as more rigorous carbon pricing, including carbon leakage protection is required to ensure a stable investment framework and to incentivize the efficient use of materials (Richstein & Neuhoff, 2022). Although CCfDs have a great potential to support clean technology innovation in energy-intensive industries such as steel industry, thereby reducing CO2, this policy only pertains to the EU member states, and not yet to Korea.



Figure 2-3 Basic functioning of CCfD

Source: Gerres & Linares, 2022

Emissions Trading Scheme (ETS)

Many studies have highlighted the importance of policies for the steel industry decarbonization, such as the Emissions Trading Scheme (ETS), as well as its link to other policies, such as CCfDs (Muslemani et al., 2021; Richstein, 2017; Richstein & Neuhoff, 2022; Vogl et al., 2021). The ETS is a market mechanism that allows countries, companies or industrial facilities that release GHGs to buy and sell their emissions as permits or allowances (Eurostat, 2023). The need for

legally-binding GHG reductions was acknowledged and agreed among participating countries in the Kyoto Protocol in 1997 (European Commission, 2015). The EU was the first in the world to present initial concepts for the ETS in 2000, then the EU ETS Directive was adopted in 2003, and the EU ETS was launched in 2005 (ibid). Korea became the first country in East Asia to launch the Korean ETS (K-ETS) in 2015, based on the Green Growth Act and the Act on Allocation and Trading of GHG Emission in Korea (GIR, 2022), and is in the third phase of the K-ETS (2021-2025) (Table 2-1). Both EU ETS and K-ETS work based on the same 'cap an trade' principle where GHG allowances are treated as a product that can be traded on the carbon market (European Commission, 2015). A cap determines the amount of GHG that can be emitted into the air by industries covered by the ETS. (ibid).

	EU ETS	K-ETS
Phase I	2005 – 2007 (3 years)	2015 – 2017 (3 years)
Phase II	2008 – 2012 (6 years)	2018 – 2020 (3 years)
Phase III	2013 – 2020 (8 years)	2021 – 2025 (5 years)
Phase IV	2021 – 2030 (10 years)	

Table 2-1 EU-ETS & K-ETS

Source: Korea Exchange, 2018

According to Vogl, Åhman and Nilsson (2021), CCfDs are co-dependent with ETS development (Vogl et al., 2021). This interdependence is particularly linked to the fate of free allocation and the respective benchmark levels that reflect the most emission-efficient installations (ibid). Allowances are allocated either by free allocation or by auctioning allowances (European Commission, 2015). The benchmark is one of the free allowance allocation measures in the ETS which determines how many free allowances an installation will receive based on its performance below a certain threshold of emissions (IEA, 2020b). To be more specific, the European Commission explains, "a benchmark is a reference value for the greenhouse gas emissions, in tCO2, relative to a production activity," (European Commission, 2015). Especially in the EU ETS, product benchmarks are determined by using "the average GHG performance of the 10% best performing installations in the EU" that produce the product (ibid). When benchmarks are used, all installations within a sector are allocated the same amount of allowances per unit activity, and the best performing installation, whose GHG emissions are lower than the benchmark, will receive more free allowances than they need (ibid). In principle, efficient facilities do not need to buy allowances, but installations that emit more than benchmark level need to buy more allowances to cover their emissions (ibid). In the case of the EU ETS, since 2013 (EU ETS phase 3), the industrial sector has received free allowances using a benchmark approach, with the benchmark level set at the average emissions of each subsector's 10% best-performing facilities (IEA, 2020b). According to the EU Commission, during phase 3, European industry associations voluntarily collected benchmark data in order to determine benchmark curves. These curves provide data points that indicate the greenhouse gas intensity per installation. The methodology used for this purpose was established by the EU commission. Based on this, the Commission selects the benchmark value by determining the average GHG emissions intensity of the 10% most efficient installations (European Commission, 2021). Benchmark values for sintered ore and hot metal are given as examples in the appendix of this paper.

If benchmarks are lowered and the number of free allocations in the sector is reduced, the recipients might get less support than anticipated (Vogl et al., 2021). The paper points out that one of the main problems is the uncertainty about the amount of free allocation if companies

shift from the blast furnace to cleaner technology and might be held to another benchmark. If a company worries that it will lose its free allocation upon switching to a low-emission production process, it will lose the incentive that carbon pricing is supposed to provide, as well as create perverse incentive that prevents companies from investing in technical innovations (ibid). Therefore, it is essential to clarify and commit to the allocation levels for development of low and zero emissions technical innovation (ibid).

Carbon Border Adjustment Mechanism (CBAM)

The CBAM is a climate measure that aims to contribute to the achievement of the EU 2050 climate target in accordance with the Paris Agreement by addressing the risk of carbon leakage³ by, for example, ensuring equivalent carbon pricing for imports and domestic products, including iron and steel (Directive 2010/75/EU of the European Parliament and of the Council of 24 November 2010 on Industrial Emissions (Integrated Pollution Prevention and Control) (Recast) (Text with EEA Relevance), 2011). CBAM is a levy on imported goods, such as steel, based on their carbon content, with the goal of balancing the costs that the EU ETS charges on European manufacturers (Vogl et al., 2021). The rationale behind the CBAM stems from the European Green Deal which outlined a path to realize the EU ambitious target of 55% reducing carbon emissions in comparison to 1990 levels by 2030 to ultimately become a climate neutral by 2050 (European Commission, n.d.-a). In line with this target, in 2021 the Commission formulated the Fit for 55⁴ policy proposals to realize this ambition (ibid), and as a part of the 'Fit for 55', the CBAM was proposed (European Union, 2023). The purpose of the CBAM is to make sure that the emissions reduction efforts of the EU are not offset by increasing emissions outside its EU borders through the relocation of production to non-EU countries especially where climate policies are less stringent and ambitious, or through increased imports of carbon-intensive products, which can undermine not only EU but global climate efforts (European Commission, 2023).

According to Vogl et al. (2021), CBAM can theoretically serve as an alternative to free allocation of emission allowances, but both schemes can be simultaneously used, which could overcompensate European manufacturers (Vogl et al., 2021). However, the paper points out that if the CBAM does not include a reverse adjustment in the return, i.e., to compensate or equalize the situation for steel that is exported from the EU to other countries, then these European steel exports would be disadvantaged. This is because it could lead to additional costs when entering non-EU markets that have their own carbon pricing mechanisms or do not have the same support system as European producers, such as CCfD (ibid).

Other demand-side market creation policies

Vogl et al. (2021) also outlined other demand-side market creation measures, including issuance of green steel certificates, labels, Green Public Procurement (GPP), quotas, etc. (Vogl et al., 2021). The paper divided market creation measures into two types: i) tradable certificates where green steel producers issue certificates that can be purchased by users in a certificate market;

³ Carbon Leakage refers to companies based in the EU moving their carbon-intensive production abroad to avoid EU regulations and to take advantage of less stringent standards, or EU products being replaced by more carbon-intensive imports (EU Commission, 2023).

⁴ The Fit for 55 package comprises a series of proposals to revise and update EU legislation and to put in place new initatives in order to align EU policies with its climate goals agreed by the Council and the EU Parliament (EU Commission, 2023). The package also aims to reform the EU ETS by increasing its level of ambition, e.g. phasing out of free aloownaces for some sectors, increasing funding for the modernisation fund and the innovation fund, etc. (ibid).

and ii) carbon footprint trancing which necessitates the collection of data at every stage of the value chain (Vogl et al., 2021). They raise concerns about the effectiveness of these measures in achieving the policy goal of supporting low-emission primary steel production. This may be attributed to an additional certificate price risk for investors, the fluctuation in the carbon price, etc (ibid). Additionally, their analysis indicates that while the implementation of these certificate schemes is viable within EU policy frameworks, such as the Ecodesign Directive, it is necessary to be extended to non-EU countries in order to adhere to World Trade Organization regulations and be applicable globally (ibid).

Green Public Procurement (GPP), for example, Vogl et al. (2021) suggests that reliable information about the embedded emissions is required to procure lower-emissions materials. GPP can be undertaken voluntarily or in conjunction with quotas or restrictions (Vogl et al., 2021). However, steel-using sectors that rely heavily on public procurement, such as construction, defense, etc., use certain steel products, e.g. rebar⁵, which are usually produced through the secondary production (Sekiguchi, 2017; Vogl et al., 2021). Therefore, GPP of green steel is not expected to significantly contribute to fostering low-emission primary production (ibid). Another study conducted by Muslemani et al. (2021), mentions GPP, as a demand-side policy, can be used as a tool by governments to set minimum green steel requirements for green steel in public procurement, or to mandate the use of a certain amount of green steel within industries (Muslemani et al., 2021). This can be implemented at industry level or applied to individual private companies operating within these industries, such as construction companies or car makers (ibid). Despite the existence of various approaches for implementing GPP, their analysis raises concerns. These include the insufficient emphasis on prioritizing economically advantageous bids that offer better life cycle performance with low GHG emissions in public tenders, and a lack of expertise within procuring authorities to develop tenders that prioritize low-carbon objectives (ibid). Moreover, in the short term, only a few green suppliers and projects that receive government support are likely to gain benefit of GPP (ibid).

Labels, for another example, which are to make externalities visible to the consumers, can support the market formation, and there are several industry labels exist nowadays such as Responsible Steel (Vogl et al., 2021). Nevertheless, due to the complexity of the steel market, market formation policy instruments could either bring new risks to deliver the full intended benefits to the producers (ibid). However, market formation policies can be supported by other policy goals, such as creating international demand for green basic materials and thereby incentivizing decarbonization efforts in other countries. The global spillovers that are created by the required effects that market formation has on the world can stimulate investment in low-emission steel production around the world (ibid). The early creation of premium markets and making information accessible could lead to an increase in the willingness to pay more for green steel (ibid).

2.2 Theoretical & analytical framework

This chapter presents the theoretical and analytical framework and ideas that underpin the study.

2.2.1 Technological Innovation Systems (TIS) framework

This study uses a theoretical and analytical framework called the Technological Innovation Systems (TIS) to examine the overall conditions for the implementation and/or

⁵ Reinforcing bar (also known as rebar) is a high-tensile steel bar, usually made from recycled steel, used in construction to increase the strength of concrete (Wood, 2023)

commercialization of the hydrogen-based steelmaking technology. The term "(national) system of innovation" was first used by Freeman (1987) and was defined as "the network of institutions in the public and private sectors whose activities and interactions initiate, import and diffuse new technologies" (Edquist, 2006; Freeman, 1987). Later, innovation systems were defined more broadly as "all important economic, social, political, organizational, institutional and other factors that influence the development, diffusion and use of innovation" (Edquist, 2006). Various researchers and policy analysts have attempted empirical studies of innovation systems to describe and understand their structure, dynamics and performance (Bergek, Jacobsson, et al., 2008). However, concerns were raised regarding the absence of comparability among these studies and the conceptual heterogeneity presenting in the literature on the innovation system. Consequently, there was a need for a practically applicable analytic framework that allows the evaluation of system performance and the identification of factors influencing performance (ibid).

The TIS framework helps to identify internal and external source of dynamics in structure and functions in a technological innovation system (Bergek, Hekkert, et al., 2008). The emergence and evolution of technological Innovation Systems (TIS) reflects the diffusion of a particular new technology. Using the TIS approach, therefore, allows to identify means of accelerating the diffusion of a technology that has been determined or influenced by entrepreneurs, policy makers and other relevant stakeholders (ibid). The framework captures the structural characteristics and dynamics of an innovation system, and the dynamics of a number of key processes - Bergek et al. referred to here as 'functions' that directly impact the development, diffusion and use of new technology, hence, the performance of the innovation system (Bergek, Jacobsson, et al., 2008). As Bergek et al. suggested, a system, in general, is defined as a group of components serving a common purpose, in other words, working towards a common objective or overall function. These components of an innovation system are the actors, networks and institutions (Carlsson & Stankiewicz, 1991) relevant to the overall function of developing, diffusing and utilizing new technology (Bergek, Jacobsson, et al., 2008). An innovation system can be primarily an analytical construct that is a tool we use to better describe and comprehend system dynamics and performance, despite the fact that the concept of a system may imply collective and coordinated action (ibid). This indicates that the system in focus may not exist as a fully-fledged entity in reality (ibid). Alternatively, it could be in the process of emergence with minimal interaction between its components, and it is also possible that the interaction among components is unplanned and unintentional, rather than deliberate even within a more advanced innovation system (ibid). These components are further explained in Chapter 4., to illustrate the context of Korea and Sweden. This research applies the framework in two ways: first, the framework helps to formulate research questions; secondly, to understand country's overall context. Berget et al. (2008) outlined seven functions and indicators in Table 1-1.

3 Research design, materials and methods

This chapter provides the research design and methodology that was used for this research as well as the processes that were carried out to construct the framework and conduct the analysis for this research to respond to research questions.

3.1 Research design

To gain a complete understanding of the challenges and opportunities associated with the adoption and scaling-up of the H-DR process in Korea and Sweden, the research design uses a qualitative, descriptive and comparative approach combined with the use of quantitative data. The research method employed in this paper is a mixed method approach, mainly qualitative research method driven with sub-quantitative analysis and data to supplement research questions and data found in the qualitative research. This approach was chosen to conduct comparative research in order to assess the strengths and weaknesses of both countries to draw lessons from each country. There have already been a several literature conducted comparative analysis using TIS framework (Bergek & Jacobsson, 2003; Vasseur et al., 2013). This literature mostly tends to focus on making comparisons across different countries to derive insights from the country that is relatively more successful in certain technologies. Inspired by the existing literature and motivated by the Swedish HYBRIT success in piloting H-DR, although the H-DR development in Sweden is more advanced than Korean, this study attempts to compare two countries on an equal footing by taking into account different contextual factors, such as differences in geography, politics, and so on to draw lessons from each country.

In order to address the first research question, i.e. "How has knowledge development and diffusion been supported in Korea VS Sweden?", this research screened through search engines such as steel producers' websites, national project search engine, the google scholar as well as reference tracing to identify the number of R&Ds on H-DR. The second and the third research questions, i.e. "How has the influence on the direction of search been supported in Korea VS Sweden?" and "How has the market formed for the green steel in Korea VS Sweden?" were mainly answered through eight semi-structured interviews and searches. These interviews used an open-ended, semi-structured approach, allowing researchers to engage with participants and elicit their perspectives freely (Creswell, 2018). To be specific, interviews with practitioners and relevant actors, such as policymakers, steel producers, etc., were necessary to capture the practitioners' perspective on the green steel transition and their opinions on the strengths and challenges of the green steel transition.

3.2 Data collection methods and materials collected

To ensure data triangulation, data for this thesis was collected by first conducting eight semistructured interviews as primary data while secondary data was gathered by examining academic and grey literature, reports, and pertinent documents using the google scholar, Korean national project search engine, etc. The collection of tertiary data included the observation of regional (the EU) and international conferences, and seminars in addition to a field visit in Luleå, Boden and Gällivare in Sweden. The observation of regional and international conferences and seminar was through HYIS forum co-hosted by POSCO and SSAB, as well as seminar was "*Decarbonization standards in the steel sector -why, what and who?*" hosted by the Jernkontoret and the Sweden Institute for Standard (SIS) (Jernkontoret & Sweden Institute for Standard, 2023). The use of data triangulation or multiple methods of data collections enhances reliability as well as internal validity of research (Merriam, 1998; Creswell, 2018). The first stage of data and material collection involved scoping this research to focus on the development of technology, H-DR, and the key actors responsible for the H-DR technical development, namely, steel producers - POSCO and HYBRIT - along with some consideration of the national dimension. In the second stage, the identification of interviewees was carried out through the network of the author's supervisor and through personally reaching out to authors of relevant literature. As illustrated in the Table 3-1, a total of ten interviews were conducted, involving five different types of institution, i.e. steel producers, NGOs, academia, Swedish steel association (known as Jernkontoret) and the Korean national think tank. Specifically, seven interviews were conducted with stakeholders from Sweden side, while the remaining four interviews from Korean side. This paper conducted four interviews from the same institution - Lund university, which is a part of the HYBRIT project, thus considered as one institution and one respondent (Respondent #8) in this research. Firstly, an email was sent to interviewes outlining the study objectives, and interview questions were sent to those who requested. Most of the interviews were conducted through the Zoom video call, and a few were conducted in person.

Country	Respondent	Organization	Department	Type of Organization
	Respondent #1	POSCO	Low-carbon iron making / HYREX	Corporate
Korea	Respondent #2	Korea Institute for Industrial Economics & Trade (KIET)	Material & Industrial Environment	Government Think tank
	Respondent #3	Korean NGO (anonymous)	Steel industry	NGO
	Respondent #4	Korea University	Energy & Environment Professsor	Academia
	Respondent #5	HYBRIT AB	HYBRIT project management	Corporate
S la r	Respondent #6	Jernkontoret	Energy, Environment and Sustainability	Iron & Steel industry's trade association
Sweden	Respondent #7	Climate Action Network (CAN) Europe	Industry policy	NGO
	Respondent #8	Lund university	Energy & Environmental system	Academia

Table	2 1	Tutomionica	for	this	mananah
1 abie	2-1	Interviewees	jor	INIS	research

Source: Author's own

3.3 Methods used to process data

The preparation of the analysis involved all the information that was obtained for this research. In terms of data collected from the interview, because of the number and the duration of interviews as well as the time constraints of this research, the transcriptions of the interviews were either fully or partially transcribed and important key notes were all included. Following this step, the data was digitized using a program called NVIVO, which is a software application for qualitative research.

4 Introducing country context

In this chapter, the overall electricity system and the background of the steel industry in each country are presented. This study does not delve further into investigating the correlation between the market system and H-DR. However, in order to provide a comprehensive understanding of the advantages and disadvantages associated with electrification for EAFs and hydrogen production in both countries, brief introduction of electricity supply and its market structure is introduced in this chapter with more focus on Korea. This chapter also provides components of the H-DR innovation system – actors, networks and institutions (Carlsson & Stankiewicz, 1991) – to see how these components contribute to the overall functions of development, diffusion and exploitation (Bergek, Jacobsson, et al., 2008) of the hydrogen-based steelmaking in each country. The scope for identifying the components is primarily national and regional context. This chapter does not consider aspects beyond the national level but mentions some relevant aspects where appropriate. The use of the concept of an "overall function" does not necessarily suggest that all actors in a particular system are present to serve that function or are directed by that function (ibid).

4.1 Electricity supply

Korea is a country with resource scarcity, but it is the world's 8th largest energy consumer, which means that it is around 94.8% dependent on imported energy (Ministry of Foreign Affairs, n.d.). Unlike European countries, Korea is geopolitically isolated, making it is difficult to export and import electricity from neighboring countries (IEA, 2021). Given this unfavorable circumstance, the electricity security has been the country's major concern (IEA, 2021).

The Korean Ministry of Trade, Industry and Energy (hereinafter, MOTIE) is required by Article 25 of the Electric Utility Act to submit a master plan to stabilize electricity supply and demand (Ministry of Government Legislation, 2021). The master plan lays out matters related to the basic direction and the long-term outlook of electricity supply and demand, plans for generation installations, major transmission and substation facilities, managing electricity demand, the expansion of distribution sources and so on. MOTIE has recently published the 10th Basic Plan for Long-term Electricity Supply and Demand 2022 – 2036 (hereinafter, the 10th BPLE), which is a mid-to long-term plan that forecasts electricity demand and designs electricity facilities and power supply configuration (ibid). According to the 10th BPLE (2023), the total installed capacity at the end of 2021 was 134 GW, LNG being the largest (31%), followed by coal (28%), renewables (19%) and nuclear (17%). And electricity generation in 2021 was around 577 TWh, while the consumption was 553.4 TWh, with industrial consumers accounting for more than 50% (MOTIE, 2023a). The electricity generation by source consists of coal (34%), LNG (29%), nuclear (27%) and renewables (7%) (ibid).

Sweden, on the other hand, the generated electricity in 2021 was around 170 TWh, consisting of hydropower (42%), nuclear (31.3%), wind (16.1%), biofuel (5.7%), waste (3%) (IEA, 2022b). Electricity generation in Sweden mainly comes from hydropower and nuclear power, however, wind power has increased significantly over the last ten years (Swedish Energy Agency, 2022). The industry sector accounts for the highest electricity consumer at 39% (47 TWh) of the total, followed by households at 35% and the commercial sector at 22% (IEA, 2019). In terms of total installed generating capacity in 2017 was 39 GW, with hydropower being the largest (41%), followed by nuclear (23%), combustible fuels (19%), wind (17%) and solar (1%) (ibid).

4.1.1 Electricity market structure

According to the many studies, a key determinant for the future competitiveness of fossil-free hydrogen steel is the cost of electricity (Åhman et al., 2018; Vogl et al., 2021; Kim & Sohn, 2022). Electricity prices can be shaped by many aspects such as electricity market structure (EDF, 2016; Strbac & Wolak, 2017). Therefore, understanding the complex relationship between the electricity market structure and electricity price is an important factor for hydrogen production and H-DR scale-up, and requires in-depth studies. In order to provide the groundwork for future research, this chapter presents a brief overview of the electricity market system in Korea and Sweden and does not discuss the relationship between the market and the H-DR development.

The Korean electricity market exhibits a day-ahead wholesale market operated by Korea Power Exchange (KPX), and by a regulated monopoly - Korea Electric Power Corporation (KEPCO) - that manages the transmission, distribution and retail of electricity to the end users (IEA, 2021). The power generation unit was liberalized in 2001 and KPX was established to install and operate the electric power market as well as the Electricity commission regulating body (Korea Energy Economics Institute, n.d.). The power generation unit comprises of six subsidiaries of KEPCO and numerous Independent Power Producers (IPPs) (IEA, 2021). Among IPPs, twenty companies operate 21 GW of installed capacity including POSCO, the first and the largest IPP in Korea (POSCO International, n.d.), and 3442 renewable energy producers registered 5 GW (KPX, 2020; IEA & KEEI, 2022). It is true that a regulated market system can help stabilize electricity supply, however, in the long-term, a liberalized market allows the risk to be shifted from end users to investors and creates incentives to operate the grid more efficiently, creating a constant pressure to drive down costs and encouraging innovation in lowcarbon technologies to enter the market (IEA, 2021). In Korea, the carbon price is not integrated into the wholesale market price and the electricity market is not conducive to lowemission technologies (IEA & KEEI, 2022).



Figure 4-1 Korea's electricity market structure

Source: KEPCO (n.d.) and IEA (2021)

On the other hand, according to the IEA (2019), the Swedish electricity market, was deregulated in 1996 and is seen as a role model for the market liberalization and regional integration (IEA, 2019). The Swedish electricity market has undergone complete liberalization, allowing customers to choose their own supplier (ibid). The separation of electricity transmission and supply activities is through unbundling and distribution network operators are functionally unbundled (ibid). The Swedish wholesale power market is part of the integrated Nordic power market with around 170 distribution network companies (European Commission, 2014). It is common for production, distribution and trading to be carried out by the same group of companies but unbundled in different legal entities (ibid). A few large generators in Sweden dominantly produce electricity including Vattenfall, a state-owned being the largest, which generates around 40% of the total. Despite a small number of generators, the Swedish Energy Markets Inspectorate (SEMI) considers the competition in the generation as competitive as of the well-interconnected Nordic market (ibid). Around 95% of the physical power trade in Sweden takes place at the Nord Pool, a regional power exchange that includes Sweden with its neighboring countries: Norway, Finland, Denmark, Estonia Latvia and Lithuania. The Nord Pool has two power markets: a day-ahead market (Elspot) and a intraday market (Elbas) (ibid). Though the intraday market is growing, the trading predominantly takes place in the day-ahead market. Currently, the Nord Pool is the only power exchange operating within the Nordic region. According to the European Union's network codes, additional power exchanges are allowed, therefore, the power exchange operation may be developed within the Nordic area (ibid). The Nord Pool day-ahead price has varied between 20 and 40 EUR/MWh, and price changes are usually caused by the availability of hydropower which is the most abundant source for large-scale generation in the Nord Pool (ibid). In terms of retail market, there are more than 120 suppliers that serve around 5.4 million Swedish end-user customers. At the end of 2017, the largest suppliers were Vattenfall, E.ON and Fortum.



Figure 4-2 Sweden electricity market structure

Source: Author's own based on IEA (2019)
4.1.2 Hydrogen

The H-DR process is based on direct reduction of iron ore using fossil free energy and hydrogen gas (Vattenfall, 2019). To give a concise understanding of hydrogen production methods, hydrogen can be produced from several sources including: reforming fossil fuels - e.g. natural gas, coal; biomass; and electrolysis from electricity that splits water (Fossil Free Sweden, 2021). Given that the H-DR method employs electrolysis, there are three main different electrolysis technologies for the production of hydrogen from water: alkaline electrolysis (ALK); electrolysis with proton exchange membranes (PEM); and high temperature electrolysis (SOEC). The first two are currently available on an industrial scale in the commercial market (ibid).

Korea

The Korean government announced its first hydrogen economy roadmap in 2019 in parallel to its national net zero target by 2050 to accelerate the decarbonization of the energy system and encourage the hydrogen economy (Korean Government, 2019). The 1st basic plan for hydrogen economy was announced in accordance with Article 5 on the Promotion of Hydrogen Economy and Hydrogen Safety Management Act that was enacted in 2020 (ibid). However, its plan was mainly focused on developing hydrogen fuel cells and transport. The roadmap outlined its plan to establish a large-scale green hydrogen production of 250,000 tons by 2030 at a unit production cost of KRW 3500/kg (equivalent to USD \$2.25/kg) and 300,000 tons by 2050 at KRW 2500/kg (equivalent to USD 1.87/kg) (See Table 4-2.) (Kang, 2022). Additionally, in the latest 10th BPLE, the Korean government plans to generate 13.0 TWh of hydrogen ammonia which accounts for 2.1% of the total energy mix by 2030 and increased to 47.4 TWh (7.1%) by 2036 (MOTIE, 2023a). Both national plans - the Hydrogen Roadmap and the 10th BPLE outline various hydrogen technologies including ammonia co-firing, blue hydrogen and green hydrogen. In terms of hydrogen-ammonia co-firing, LNG and coal will be co-fired with hydrogen and ammonia to produce 6.1 TWh by 2030 and 6.9 TWh by 2036 respectively (ibid). Korea is considered as a fast mover in terms of officially developing the national hydrogen strategy (Hong & Kim, 2022), however, there are some shortcomings. One of main challenges is the technology gap. The average electrolysis efficiency of overseas companies is around 60% and is at the demonstration stage at the MW level while that of Korea is at 55% at the KW level (See Table 4-1) (MOTIE, 2021).

Part of POSCO's business strategy is to produce a massive amount of green hydrogen with the aim of 5 million ton of hydrogen production system by 2050 (Newsway, 2020; POSCO, 2021b). POSCO has started cooperating with many international companies, such as FMG – an Australian iron ore company – for green hydrogen business, and Orsted – a Danish offshore wind power company – to supply steel materials for the construction of an offshore wind farm as well as green hydrogen production. Furthermore, it is noteworthy that POSCO announced its business agreement with the Korean Research Institute of Industrial Science & Technology (RIST⁶) for hydrogen in 2021 (ibid). In January 2023 the Korean government has set a target of KRW 20 trillion in green industry export orders, and has formed a public-private partnership including POSCO, two subsidiaries of KEPCO⁷, Samsung and etc. (Ministry of Environment, 2023). One of plans in this target includes green hydrogen production and seawater desalination with the aim of exporting a total value of KRW 17.5 trillion (equivalent to EUR 12.3 billion and USD 13.2 billion) to the Middle East including Oman, Saudi Arabia, and the United Arab Emirates (ibid). And in June 2023, this partnership for green hydrogen export plan successfully

⁶ A private research institute specialized in technology and commercialization that was founded by POSCO (RIST, n.d.)

⁷ Korea Electric Power Corporation (KEPCO) is the largest state-owned electric utility in Korea and responsible for the generation, transmission and distribution of electricity (KEPCO, n.d.)

led MOU contract between the Korean ministry of Environment and the Omani ministry of energy and minerals (Chosun, 2023). It is believed that the Omani government's interest in green hydrogen stemmed from its recognition of the importance of realizing its phase-out of fossil fuels (ibid). It was important for the Omani government to secure demand for its green hydrogen production, while the Korean government also needed to establish the green hydrogen supply chain, thus their interests were converged (ibid). The project entails the production of green hydrogen from wind and solar energy with an annual production capacity of 1.2 million tons of the form of green ammonia on a 320km2 site in Duqm, Oman which is the largest scale in the world (Ministry of Environment, 2023a). According to IEA, Oman has a potential to produce green hydrogen at the lowest cost of USD 1.6/kg by 2030 (IEA, 2023). POSCO Holdings is the main shareholder with a 28% ownership followed by the ENGIE - a French energy corporate (25%), and Samsung Engineering (12%), etc. (Hankyung, 2023).

Table 4-1 Green hydrogen production target comparison

	Korea (2028)	Sweden (2030)	Germany (2030)
Green hydrogen production target	10 MW	3 GW	5GW

Source: MOTIE (2021) & Fossil Free Sweden (2021)

Table 4-2 Hydrogen price target in Korea and Sweden

Country	2030	2050
Korea (unit: KRW)	3500/kg (eq. EUR 2.12 or USD 2.25)	2500/kg (eq. EUR 1.77 or USD 1.87)
Sweden (unit: SEK)	10-11.5/kg (eq. EUR 0.9 or USD 1)	5.5 – 6/kg (eq. EUR 0.5 or USD 0.52)

Source: KEEI (2022) & Fossil Free Sweden (2021)

Sweden

Sweden put in place the government initiative, "Fossil Free Sweden", a platform for government and industry collaboration, and published its hydrogen strategy in January 2021 (Green Hydrogen Organization, n.d.). The main focus of the strategy is to decarbonize the country's energy-intensive industries including steel production with a target of 3 GW by 2030 (Table 4-1) (Fossil Free Sweden, 2021). Later, the Swedish Energy Agency suggested a proposal for a national fossil-free hydrogen strategy to stop climate change and reduce the country's reliance on fossil fuels (Green Hydrogen Organisation, n.d.). This proposal has a more ambitious target for hydrogen production: 22-42 TWh of green hydrogen by 2030 and 44-84 TWh by 2045 as well as a total electrolyser capacity of 5GW by 2030 and 15 GW by 2045 (Swedish Energy Agency, 2021b). It is important to note that the growing interest in fossil-free hydrogen in Sweden is largely motivated by the government's goals of achieving climate neutrality by 2045 with industry's collaborative efforts towards the same objective (Cetković & Stockburger, 2023). This goals are articulated within the Climate Policy Framework, which received substantial support in the Swedish parliament in 2017 (Ministry of Climate and Enterprise, 2021; Cetković & Stockburger, 2023). Fossil-free hydrogen is considered as crucial to the decarbonization of the Swedish energy-intensive industries and in securing Swedish competitiveness in the future global low-carbon economy. Hydrogen is also viewed as a viable tool to help balance the electricity grid in light of the increasing share of wind power (Ćetković & Stockburger, 2023). Overall, this highlights a unique feature of Sweden that the internal partnership between the Swedish government and corporates is robust.

4.2 Korean & Swedish Steel Industry

This section illustrates the overall Korean & Swedish steel industry shedding light on POSCO and HYBRIT with emphasis on unique features in addition to blast furnaces and the H-DR process.

Steel industry in Korea

The steel industry in Korea began in the mid-1960s. The Korean government selected Pohang as the location for the integrated steel mill in 1967 and in the following year, Pohang Iron and Steel Company was founded (POSCO, 2018). POSCO operates two integrated steel mills in Korea with a total of 8 blast furnaces that are currently operating – 3 in Pohang, 5 in Gwangyang - and 2 FINEX (Figure 4-3) (Global Energy Monitor, 2022a, 2022b). The blast furnace in Gwangyang has the biggest production capacity in the world with its internal capacity of 6000m3 as the Table 4-3 shows (POSCO, 2019). The crude steel production in 2021 was 42.96 million tons (World Steel Association, 2022). The by-product gases, typically carbon monoxide, CO2 and nitrogen, that produced inside the blast furnace are emitted about 1600 m3 of gas, and POSCO recycles this by-product gas and self-produces 74% of the electricity for its own steelmaking operation (POSCO, 2019).



Figure 4-3 POSCO integrated steel mills locations

Source: Author's own created in qGIS

POSCO has a unique steelmaking technology called FINEX that has been brought online since 2007 (Yi et al., 2019). The FINEX process was initiated by POSCO and Primetals in 1992 to create an alternative process to the blast furnace by eliminating the coking and sintering processes and using lower grade, lower cost raw materials (ibid). POSCO went through a series of scale-up development from a 15 ton/day model plant started in 1996 to a 150 ton/day pilot plant in 1999 followed by a 0.6 MTPA⁸ FINEX demonstration plant in 2003 (ibid). The first commercial FINEX plant with a capacity of 1.5 MTPA and the second commercial plant with a capacity of 2 MTPA were implemented in Pohang. FINEX consists of four-unit processes: a multi-stage fluidized-bed reactor, hot-compacters, coal-briquettes and a melter-gasifier (ibid).

⁸ Million Ton per Annum

Simply put, the biggest strength of the FINEX is the direct use of iron ore fines and non-coking coal while eliminating the coke-making and sintering processes that are the precursor to the blast furnace (POSCO, 2017b). With this experience of developing the FINEX process, POSCO and Primetals are again collaborating to build HYREX, which share same process method but replacing non-coking coal to hydrogen (POSCO, 2022d).



Figure 4-4 BF and FINEX process comparison

Source: POSCO & Primetals, (n.d.)

Table 4-3 POSCO plants in Pohang and Gwangyang

Location	Туре	Start (Closing year)	Capacity	Size (m [*])	Crude Steel Production (2021)
	BF 2 (BOF)	1987 - 2025	5 MTPA	5600	
	BF 3 (BOF)	1978 – 2025	5 MTPA	5600	16.9 MTPA
Pohang	BF 4 (BOF)	2010 - 2025	5 MTPA	5600	
	BF-DRI	1995 -	0.8 MTPA	No information	
	FINEX No. 2	2007 -	1.5 MTPA	No information	No information
	FINEX No. 3	2014 -	2.0 MTPA	No information	No information

	BF 1	1987 -	5.65 MTPA	6000	
	BF 4	1992 -	5 MTPA	5500	
Gwangyang	BF 5	2000 -	5 MTPA	5500	23 MTPA
	BF 3	1990 -	4 MTPA	5500	
	BF 2	1988 -	4.5 MTPA	3685	

Source: Global Energy Monitor Wiki (2022)

Steel industry in Sweden

Swedish iron and steel production has a long history. Since industrialization, the Swedish steel industry has grown significantly throughout the 19th century having over 200 blast furnaces (Jernkontoret, 2018). Swedish steel products are small in volume but known for their niche steel. The history of SSAB goes back to the 1970s and 1980s, when the Swedish steel industry worked to improve its efficiency, partly driven by the state. And these activities led to the naissance of SSAB, where the production of commercial steel was concentrated in one company, initially with the Swedish state as half owner. In the 1970s, the export share was 40-50%, but in 2019 it stands around 85% (ibid). Some of Sweden's comparative advantages are: good availability of raw materials such as iron ore; cheap and stable electricity supply; abundant water; and highly educated population in metallurgical research (ibid). Crude steel production in Sweden amounted to 8.8 million tons in 2021 (World Steel Association, 2022). Sweden has two active blast furnaces in Luleå and Oxelösund, both owned by SSAB, while HYBRIT's H-DR pilot plant is located in Luleå, close to SSAB's blast furnace, and its demonstration plant is planned to be located in LKAB's industrial area in Gällivare (Figure 4-5) (HYBRIT, 2022a).



Figure 4-5 HYBRIT's H-DR facilities locations

Source: Author's own created in qGIS

Table 4-4 SSAB's B	F plants
--------------------	----------

Location	Type & Unit name	Start (Closing year)	Capacity	Size (m ³)	Crude Steel Production
Luleå	BF 3	2000	2.5 MTPA	2540 m³	2.3 MTPA (
Oxelösund	BF 2	1952 - 2025	0.5 MTPA	760 m ³	1.5 MTPA

Source: Global Energy Monitor Wiki (2022)

It is unclear how the three companies found each other and initiated the HYBRIT project. None of the identified literature clearly stated what motivated them to collaborate. However, what was clear is that this project started in 2016, after the Paris Agreement. Sweden's environmental leadership was already well-known even before the Paris Accord. Before the Paris Accord, the Swedish government committed to take action and to make Sweden the first fossil-free welfare state in the world moving even faster than the EU regulations (MacDonald, 2015). The Paris Agreement, where countries agreed on a new global climate strategy that lays out the basis for future long-term climate efforts, was the frosting on the cake for Sweden's environmentally conscious leadership. According to Martin Pei - the Executive Vice President at SSAB, et al., to make the steel industry the Paris Agreement compatible, 400-600 million tons of CO2 per year needs to be decreased compared to that of 2015 which is 2800 million tons CO2 (Pei et al., 2020). In response, the Swedish parliament passed a climate law with net zero target by 2045. Soon after April 2016, SSAB, LKAB and Vattenfall AB initiated the HYBRIT project conducting a pre-feasibility study with support from the SEA (ibid). Therefore, the Paris Agreement has prompted environmentally conscious leadership in three companies (or one of three companies) to initiate the project.

4.3 TIS Components

According to Bergek et al. (2008), "the components of an innovation system are the actors, networks and institutions contributing to the whole function of developing, diffusing and utilizing new products (goods and services) and processes" (Bergek, Jacobsson, et al., 2008). Therefore, this chapter presents TIS components of the H-DR technology in each country.

4.3.1 Actors

In order to identify and analyze the structural components of the system in Korea and Sweden, it is first necessary to identify the actors of the TIS. This encompasses the entire value chain, including universities, research institutes, public bodies, direct and indirect interest groups and so forth.

Numerous actors are engaged in hydrogen-based steelmaking in Korea. The major steelmakers setting the carbon neutral strategy involving hydrogen-based iron/steelmaking are POSCO and Hyundai Steel. Several Korean government bodies that are in charge of technology development are involved in the steel industry decarbonization (Respondent #1). There are some public research institutes such as the Korea Energy Economics Institute (KEEI) and Korean institute for Industrial Economics & Trade (KIET) that are actively engaged in releasing policy briefings,

reports, etc. H-DR iron and steelmaking technology development has become a national project, thus, many universities have joined and become part of the project, such as Seoul National University, Korea university, Yeonsei university, Pohang University of Science and Technology, POSCO Engineering & Construction (E&C), Chosun Refractories Co., Ltd., etc. (Respondent #1 & #3).

Sweden has a large number of actors that are active in the hydrogen-based steelmaking. The main companies involved in the hydrogen-based steelmaking technology are Hydrogen Breakthrough Iron making Technology (HYBRIT) - a joint project together with SSAB, LKAB and Vattenfall - and H2 Green Steel (H2GS). The demand side is comprised of mostly automobile companies such as Volvo AB, BMW, etc. (H2GS, 2022; HYBRIT, 2021a). The public body that are engaged in steel space is Swedish Energy Agency (SEA) and the Ministry of Climate and Enterprise. The SEA is engaged in facilitating of energy efficiency measures and investments in renewable energy technologies (SEA, n.d.). The ministry is responsible for the climate, environment, energy, enterprise, innovation, and transition to a circular economy (Regeringskansliet, 2014). The Swedish parliament also an active actor in steel space setting up a committee of how to reach a climate target and the committee organized a series of roundtable discussion in terms of heavy industry, including steel, where industry and researchers discuss. The HYBRIT actively collaborates with academia and their names such as Lund university, and field of responsibility in the HYBRIT project was clearly stated. Jernkontoret, an industry institute organization, a.k.a. iron office, organizes the Sweden steel industry, plays a role as a bridge organization that links politics and industry (Respondent #8).

4.3.2 Network

The second structural component of interest is that of networks. Both formal and informal (Bergek et al., 2008). Steel is a commodity that depends heavily on global trading, therefore, the network within the steel industry is strong. Moreover, Sweden and Korea, where there was no strong interface that draws these countries together in terms of trading, nonetheless, because of the common decarbonization aim transitioning to the hydrogen-based steelmaking, organized a global annual forum that brings all relevant actors including academia, hydrogen producers, energy producers, international organization, etc. in the steel space.

In the case of Sweden, knowledge networks between the three shareholders of the HYBRIT (SSAB, LKAB and Vattenfall), public bodies (Swedish Energy Agency and RISE Research Institute of Sweden), and academic research groups (Lund university, KTH, Lulea university of Technology, Sandvik⁹, have been actively engaged in the research project carried out by HYBRIT and contributed to knowledge formation and diffusion (HYBRIT, n.d.). Not many actors from non-governmental organizations (NGOs), nor networks between NGOs and Swedish steel producers were spotted, however Climate Action Network Europe (CAN Europe) said they continue dialogue with H2GS in terms of the EU Emissions Trading Scheme (Respondent #7).

In Korea, networks happen between steel producers, e.g., POSCO, Hyundai Steel, etc., public bodies, academia, NGOs and civil society. Learning networks are considerably more prominent than other networks such as political networks. Steel producers and universities often interact in national projects. Examples of current learning networks are the national project on the hydrogen-reduction steel and the international forum, HyIS, allowing more international

⁹ Sandvik is a global tech engineering group providing solutions for the manufacturing, mining and infrastructure industries to enhance productivity, profitability and sustainability.

exchange of knowledge and experience. On the other hand, networks with the government were hard to identify.

4.3.3 Institutions

The third element that needs to be identified is institutions, which include culture, norms, regulatory aspects, cognitive rules that regulate interactions between actors, define the value base of different segments of society, influence the decision of businesses, and structure learning processes and routines (North, 1994; Bergek et al., 2008; Bergek et al., 2008a). Institutional change and its associated politics are crucial in the process of new technology becoming effective (Bergek et al., 2008). In order to accommodate and promote the adoption of technologies, the rules and systems that regulate how things work need to be adjusted (ibid). This shows that firms that compete TIS, compete market for products and services, and institutional influence (ibid). Thus, rival enterprises frequently cooperate to manipulate the institutional environment to acquire access to resources for their collective survival (Van de Ven & Garud, 1987).

The institutions have played an important role in Sweden in terms of accelerating climate action in the steel industry. On behalf of the Swedish steel companies, Jernkontoret, the Swedish Iron & steel Association (translated by the author), has released the Climate Roadmap for the steel industry in Sweden that laid out possible measures for emission reductions, policy for a competitive decarbonized steel industry, demonstrating their shared commitment (Jernkontoret, 2018). SSAB, LKAB and Vattenfall, have also independently released their respective roadmaps with all of them aiming to achieve fossil-free by 2045 in line with the national climate target (LKAB, 2022; SSAB, 2017; Vattenfall, n.d.). These individual initiatives underline Swedish steel industry's proactive commitment to achieving sustainability goals. In addition, the Swedish government has played a crucial role as an interface to bridge the industrial sector and politics to speed up the transition. For instance, the Swedish government started the Fossil Free Sweden initiative in 2015 establishing 'Fossil Free Sweden' initiative in 2015 whose goal is "to build a strong industrial sector and to create more jobs and export opportunities by going fossil free" (Fossil Free Sweden, 2021). Fossil Free Sweden collaborates with actors, i.e., companies, municipalities, regions and organizations to identify obstacles and opportunities to speed up growth. Fossil Free Sweden develops political proposals that are presented to the government and gathers stakeholders to implement measures (ibid).

In the case of Korea, a few literature on Korean steel industry were identified including POSCO's annual report, and reports that explain the overall Korea steel industry from national research institutes such as KEEI, KIET, etc., and NGOs. Despite the existence of some publications addressing various aspects of the Korean steel industry, there is a lack of an integrated and cohesive body of work that systematically scrutinizes the Korean steel industry's transition to a low-/zero-carbon economy (Respondent #1, #2 & #4). In other words, a coherent framework and structure among actors is absent, therefore, further research, networks and institutions that can bring these fragmented pieces of work and build an organic connection to comprehensively examine and support the green steel transition is a must. Recently, the Ministry of Trade, Industry and Energy (MOTIE) chaired a 'Steel Industry Development Roundtable' and announced the 'Steel industry development strategy for Transition to lowcarbon Steel Production'. The meeting was attended by seven major steel companies including POSCO and Hyundai, Korea Iron & Steel Association (KOSA), Korea Steel Scrap Association (KOSIA), Korea Institute for Industrial Economics & Trade (KIET) and Yeonsei University (MOTIE, 2023). Hopefully, this initiative can contribute to a more organically harmonized steel industry green transition. This initiative by the government holds the potential to enable the

organic harmonization of efforts among stakeholders, resulting in a more integrated approach towards attaining decarbonization with the Korean steel industry.

5 Findings

This chapter presents the findings and analysis of the research based on the TIS functions – knowledge development; direction of the search; and market formation.

5.1 Function 1 Knowledge development & diffusion

Knowledge development is crucial in the innovation system as it provides the necessary fuel for entrepreneurial experimentation and facilitates an understanding of how technological alternatives can be optimized for successful market outcomes (Vasseur et al., 2013). As the learning process frequently involves different actors, knowledge exchange emerges as another crucial aspect (ibid). This function captures the breadth and depth of the current knowledge base of the TIS, as well as its evolution over time, encompassing the diffusion and combination of knowledge within the system (Bergek, Jacobsson, et al., 2008). This study identifies the number of R&D projects, investment in R&Ds and knowledge sharing activities.

In Korea domestically, one respondent (Respondent #1) said that most of the HYREX project collaborations are with two government ministries – MOTIE and MSIT, and academia – POSTEC, Seoul National University, Yeonsei University, Korea University, Hanyang University, etc. There is one governmental research institute dedicated to steel in terms of alloy design, process and property control (Korea Institute of Materials Science, KIMS) but there are no governmental research institute that are specialized in ironmaking and steelmaking process (Respondent #1). The information on HYREX was available on POSCO's website and was mentioned in several policy briefings. Recently, POSCO has launched a website dedicated to HYREX¹⁰. However, certain information on the website is currently accessible only to its partners. Therefore, to complement and to identify the number of past and ongoing R&D projects on H-DR, two additional search engines in addition to Google scholar, were used: the National Science & Technology Information Service (NTIS) - a platform that provides information on past and ongoing national R&D (NTIS, n.d.), and the National Knowledge Information System (NKIS) – a policy information sharing platform from National Research Council and national research institutes (NKIS, n.d.).

From the NTIS, when "hydrogen direct reduced ironmaking", or "hydrogen reduction ironmaking" ("수소환원제철" in Korean) was searched, a total of 32 projects appeared. Among them, projects that included "hydrogen direct reduction ironmaking" as a keyword were extracted, in other words, nuclear hydrogen, CO2 emission reduction for Blast Furnace, scrap, etc. were excluded. It was then narrowed down to 2 projects: 1) Development of CO2 reduction type Hybrid Amplification and Reforming Technology for Reduction ironmaking (2017-2024); and 2) Fundamental study of the thermodynamics and kinetics of CO2 utilization and nitrogenation in future steelmaking for carbon neutrality (2022-2027). These two projects have produced 2 and 1 research papers¹¹ respectively and 19 patents solely from the former. The investment scale of these two R&D projects has been KRW 11.8 billion and KRW 800 million won, for a total of KRW 12.6 billion won (equivalent to approx. USD 9.6 million or EUR 9 million) (ibid). These two projects include hydrogen-based iron & steel making technology, however, it can be argued that they are more complementary components of a larger system rather than standalone topic. It was not until May 2023 that the H-DR became the main focus

¹⁰ http://www.hyrex.co.kr

¹¹ Namely Long-term durability of La0.75Sr0.25Cr0.5Mn0.5O delta-3 as a fuel electrode of solid oxide electrolysis cells for coelectrolysis ; Development of CO2 reduction type Hybrid Amplification and Reforming Technology for Reduction ironmaking; and Critical challenges facing low carbon steelmaking technology using hydrogen direct reduced iron

of the national project. The government has selected 100 core technologies, including POSCO's HYREX technology, to achieve carbon neutrality (MSIT, 2023). It is intended to first complete basic technology development by 2025 with the government's R&D investment of KRW 26.9 billion (equivalent to EUR 19 million) including KRW 16.9 billion for fluidized bed reactors and KRW 10 billion for electric smelting are furnace. Considering that it takes a long time to fully introduce HYREX, KRW 237 billion (equivalent to EUR 173 million), R&D investment on low-carbon fuel substitution and energy efficiency on current BFs and EAFs by 2030. This investment involves in KRW 141.4 billion by the government (MOTIE, 2023c). From NKIS, when H-DR was searched, there were a total of 10 reports that at least dedicated a chapter with H-DR. Among them, two out of ten publications specialized in the steel industry decarbonization. However, it is important to note that these reports constitute grey literature produced by a government research institute rather than collaborative research including corporates and other stakeholders.

In Sweden, SSAB, LKAB and Vattenfall have collaborated on the HYBRIT project with financial supports from the Swedish Energy Agency (SEA) and the EU's Innovative Fund. In early 2016, these three companies announced the implementation of a fossil-free steel production process in Sweden, and a pre-feasibility study was launched (Åhman et al., 2018; HYBRIT, n.d.). From 2016 to 2019, the HYBRIT initiative has received significant financial support of SEK 599 million (equivalent to EUR 51.5 million) from the SEA when initiating the project (Vattenfall, 2019). In 2017, a collaborative research program between industry, universities, and research institutes was established to identify possible fossil-free stages in the energy, mining, iron and steel value chain (ibid). Consequently, knowledge development on the hydrogen-direct reduced ironmaking technology has been able to be fostered by many different actors. The HYBRIT initiative has divided the work packages that is necessary for the HYBRIT project. Table 5-1 shows the HYBRIT work packages with different partners and their published research papers (HYBRIT, n.d.-a). According to Gerres et al. (n.d.), although Swedish companies have smaller steel production capacity compared to Korean steel producers and other major EU steel companies, Swedish companies have announced the largest share of investments, with a South Korean company coming in second (Table 5-3) (Gerres et al., n.d.).

Work package	Work package's partners	Number o papers	f
Supply of fossil-free electricity and the effects on the electricity system	LKAB, Luleå University of Technology, Swerim and RISE ETC with the support of Vattenfall and SSAB		
Production and storage of hydrogen	Vattenfall and KTH	12	
Production of fossil-free iron ore pellets	LKAB, Luleå University of Technology, Swerim and RISE ETC with the support of Vattenfall and SSAB	7	
Hydrogen-based reduction of iron ore	LKAB, KTH, Luleå University of Technology with the support of Swerim and RISE ETC	6	

Table 5-1 HYBRIT work packages and partners

Steel production of hydrogen-reduced iron ore	by SSAB and KTH with support from LKAB and Luleå University of Technology	4
Meeting the market	Lund University, Stockholm Environment Institute (SEI) and Swerim with support from SSAB, LKAB and Vattenfall	15

Source: Hybrit (n.d)

	Table 5-2	Korean	national	project	for	H-DR	from	NKIS
--	-----------	--------	----------	---------	-----	------	------	------

	Title	Organization	Year	Keywords & Summary
1	Scenario analysis of iron and steel production process for carbon neutrality	Korea Energy Economics Institute, Inha University	2022	Analysis on the cost of fuel and raw material of BF and H-DR
2	Assessment of Carbon Neutrality Potentials from Sector Coupling	Korea Energy Economics Institute, Pukyeong National University	2022	P2G technology: converting energy surplus into hydrogen
3	A study on Policies Supporting Decarbonization in Hard-to- Abate Industry	Korea Energy Economics Institute, Busan National University	2021	hard-to-abate sector, low-emission practice, policy support
5	Energy Transition Forum for Green New Deal	Korea Environment Institute	2020	Green New Deal, Industrial transition, Clean energy transition
6	Costs of Innovative Technologies for Achieving Carbon Neutrality	Korea Environment Institute	2022	Carbon Neutrality, Technology costs (EES, Hydrogen, etc.)
7	A study on Changes in Trade and Environmental Relations and Korea's Response Strategy in the Era of Carbon Trade	Korea Environment Institute, BNZ Partners, Korea University	2022	СВАМ
8	A study on how to facilitate GHG emissions decoupling in major manufacturing industries	Korea Energy Economics Institute	2020	hard-to-abate sector, decoupling
9	A study on the strategies for early settlement of market driven hydrogen economy in Korea	Korea Energy Economics Institute	2021	Hydrogen economy

Source: National Knowledge & Information System (n.d.)

Numerous universities participate in national projects related to steel. However, Respondent #1 expressed that there is a shortage of master's and PhD students in the field of iron and steelmaking process as steel is considered to be declining industry compared to semiconductor, AI and other high-tech industries. Research programs or research papers on H-DR steelmaking technology from Korean steel producers specifically was not identified neither on the NTIS

platform nor the POSCO's website. Although H-DR specific R&D collaborations are not pronounced, Respondent #4 said that there is active R&D collaboration between government & academia on green hydrogen (Respondent #4).

Although it was difficult to find much active domestic knowledge development within Korea, e.g. research publications, projects etc., POSCO has actively sought cooperation internationally. In 2021, POSCO signed a Memorandum of Understanding (MoU) with Roy Hill¹² – Australia's largest iron ore mine operator, producing 60 million tons of iron ore per year (Yeonhap, 2021). The aim was to conduct a preliminary feasibility study on an optimal hot briquetted iron (HBI) production system combining POSCO's hydrogen reduction steel technology (Roy Hill, 2022). In March 2022, POSCO Group made a Heads of Agreement with Australia's Hancock group on a joint study on HBI, with plans to invest in Australia in renewable energy and water electrolysis for hydrogen production and to produce HBI (Hankyung, 2022; Sparke, 2022). In addition, POSCO has launched the first international Hydrogen Iron & Steel Making Forum (HyIS) since 2021, which has become an annual international event providing a platform for stakeholders, including steel producers, electricity utility, hydrogen producers, international organizations, steel buyers in steel industry to gather and share knowledge on decarbonized hydrogen steelmaking. Such an initiative has fostered knowledge development on a global level.

Table	5-3	Numl	per of	research	and	the	investment
			./				

	The total number of Research	Total amount of initial R&D
	paper	investment
Korea	26	EUR 8.2 billion
Sweden	43	EUR 20 - 46 billion

Source: National Knowledge & Information System (n.d.) & Gerres et al., n.d.

5.2 Function 2 Influence on the direction of search

For the development of a TIS, a diverse array of enterprises and other organizations must choose to enter it (Bergek et al., 2008). Consequently, it is imperative to provide sufficient incentives and/or pressures to motivate organizations to be induced to do so (ibid). The second function covers the mechanisms that impact the direction of search within the TIS, including but not limited to different competing technologies, applications, markets, business models, etc. These factors are, for instance: incentives from changing factor and product prices, growth occurring in TISs in other countries, actors' perception of the relevance of different types and sources of knowledge, regulations and policy, articulation of demand from leading customers, etc. (ibid). Among these factors, this paper investigates more into regulations and policy for the steel industry in Korea and Sweden.

Based on the interviews carried out with Sweden/EU respondents, two most frequently mentioned policies were Industrial Emissions Directive (IED) and Emission Trading Scheme (ETS). Additionally, one respondent highlighted the Ecodesign Directive as well (Respondent #7). The Korean perspective mostly emphasized the Carbon Border Adjustment Mechanism

¹² Roy Hill is an independent iron ore operator in Western Australia's Pilbara region with four ownerships: Hancock Prospecting Pty Ltd (70%), Marubeni Corporation (15%), POSCO (12.5%) and China Steel Corporation (2.5%) (Roy Hill, n.d.)

(CBAM) and carbon pricing. In addition to CBAM, the Korean respondents did not specify a particular policy, but one respondent said that "Korea has many policies, but they are rather scattered and there is a lack of integrity" (Respondent #2).

5.2.1 Investment

In terms of investment in Korea, MOTIE announced a development strategy for the steel industry which laid out its plan on how to secure & strength scrap and R&D on low-carbon steel technology development including hydrogen reduction technology. Along with this, steel decarbonization alliance will be formed to link public-private partnerships and establish KRW 150 billion private fund (equivalent to EUR 107 million) to create scrap value chain and low-carbon iron & steel making transition (MOTIE, 2023).

Numerous articles have been published regarding the investment to Swedish hydrogen-based steel production. In 2019, the three shareholders of HYBRIT – SSAB, LKAB and Vattenfall – announced investing SEK 150 million and the SEA is contributing additional 50 million for the construction of a storage facility for hydrogen at the HYBRIT pilot plant (approx. EUR 17.3 million) (MaintWorld, 2019). In 2022, the HYBRIT was granted a total of EUR 143 million for scaling up for a complete value chain for hydrogen-based iron and steelmaking from the EU Innovation Fund¹³ (HYBRIT, 2022). The grant consists of EUR 108 million earmarked for the demonstration of the hydrogen direct reduction process which involves fossil-free hydrogen production in Gällivare (HYBRIT), another EUR 30 million for the demonstration of electric melting of hydrogen-based direct reduced iron in Oxelösund (SSAB) and the last EUR 5 million for the demonstration of fossil-free DR-pellets production for the hydrogen reduction process (LKAB) (ibid).

5.2.2 Industrial Emissions Directive (IED)

The IED serves as the EU instrument regulating pollutant emissions from industrial activities (European Commission, n.d.). The IED encompasses a range of emissions, including emissions to air, water and land (originally excluding CO2 emission), generation of waste, use of raw materials, energy efficiency, noise, accident prevention, and site restoration after closure. Approximately 52,000 installations in EU – 1300 plants in Sweden (Jernkontoret, 2021) - that engage in the industrial activities are listed in Annex I¹⁴ of the Directive and required to operate under a permit granted by the authorities in the Member States (European Commission, n.d.). The permit should contain conditions set in accordance with the principle of the 'polluter pays' and of pollution prevention (Directive 2010/75). The permit conditions, including emission limit values, are established based on the Best Available Techniques (BAT). The commission facilitates an information exchange with experts from Member States, industry and environmental organizations to define BAT and its associated environmental performance within the EU context (European Commission, n.d.)¹⁵. The outcome of this process yields BAT

¹³ The EU Innovation Fund is one of the world's largest funding programs that supports the demonstration of innovative lowcarbon tehnologies that mitigate reduce GHG emissions (HYBRIT, 2022).

¹⁴ Iron and Steel production is listed in the Annex I chapter 2. "Production and processing of metals".(Directive 2010/75/EU of the European Parliament and of the Council of 24 November 2010 on Industrial Emissions (Integrated Pollution Prevention and Control) (Recast) (Text with EEA Relevance), 2011) https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A02010L0075-20110106

¹⁵ This process is coordinated by the European Integrated Pollution Prevention and Control (IPPC) Bureau at the EU Joint Research Center in Seville, Spain (European Commission, n.d.)

Reference Documents (BREFs)¹⁶ which contain the BAT conclusions (BATC)¹⁷ that are subsequently adopted by the Commission as Implementing Decisions. The IED requires that these BAT conclusions are the reference for setting permit conditions (ibid).

The steel industry is known as the first sector to receive BAT Conclusion for iron and steel production, which means that the steel producing companies have been able to meet the emission limit value based on the BAT specified in the BAT Conclusions (Jernkontoret, 2021). The activities of the Swedish steel industry are regulated by IED, the associated descriptions of BAT and emissions limit values to be achieved (Jernkontoret, 2018). During the years 2020-2023, the IED goes through a revision in line with the European Green Deal (ibid). Some of the main proposals for revision of the Directive are: i) to extend it to cover more processes and new sectors (e.g. whether mining should be included); and ii) to include GHG emissions reduction in installation-level environmental permits, and so forth. (CAN Europe, 2022; European Commission, n.d.-b; Jernkontoret, 2021). One respondent expressed concern regarding the IED revision. This is because the Commission and other legislators want to widen and put more details into IED which may result in challenges in obtaining permits for specific steel production process. To be specific, one concern is the risk of overlapping with other legislation, such as EU Emissions Trading Scheme (ETS) and waste legislation. Another concern is the inclusion of CO2 in the IED. The steel industry is included within the EU ETS. As per the provisions of Article 9.1 of the IED, it is stipulated that the permit shall not include an emission limit value pertaining to the direct emissions of GHG (Jernkontoret, 2021). The rationale behind the IED not including CO2 was that CO2 would be already covered by the ETS. However, because of the issue of free and over allocation from the EU ETS, the CO2 emissions reduction may not be effectively achieved, as indicated by one of respondents (Respondent #7).

5.2.3 Emissions Trading System (ETS)

The Kyoto Protocol to the UN Framework Convention for Climate Change (UNFCCC) was agreed upon in 1997 and set legally binding GHG reduction targets by 2012, for 37 industrialized countries (Annex I countries) during the first commitment period (2008-2012) (European Union, 2015). This necessitated the development of policy instruments to fulfil the Kyoto commitment (ibid). Based on this need, the European Commissions presented a green paper outlining initial concepts for the EU ETS in 2000 (European Commission, n.d.). It served as the foundation for many stakeholder discussions that contributed to the system's development (ibid). The EU ETS Directive then was adopted in 2003 and the EU ETS was launched in 2005, the world's first international emissions trading system (European Commission, n.d.). In conjunction with the GHG emissions inclusion,

Under the EU ETS phase 3 onwards, a benchmarking approach is used for the free allocation of allowances (European Commission, n.d.). The free allocation for each installation is calculated using GHG emission benchmarks developed for each product, where possible and 54 benchmarks have been currently elaborated based on extensive technical work and consultations with various stakeholders. A product benchmark is based on the average GHG

¹⁶ BREFs refer to what is considered to be the best available techniques for a sector or part of a sector (Jernkontoret, 2021).

¹⁷ The BATC is a chapter in BREFs. It outlines the BAT and associated emission levels (BAT-AELs) for different activities under the IED (ibid)

emissions of the best performing 10% of the installations producing that product in the EU and EEA-EFTA states (European Commission, n.d.).

The benchmark is used to determine the level of free allocation that each installation in each sector will receive. A benchmark is not an emission limit or an emissions reduction target per se (European Commission, 2015). All installations within a sector receive the same allocation of allowances per unit activity. The best performing installation, whose GHG emissions are lower than the benchmark, will receive more free allowances than they need (ibid). In other words, the principle for these benchmarks in the beginning was to look into all the sites making the same product and their emission intensity and then take the 10% best of producers in terms of having the cleanest and least-polluting facilities in each sector. These selected benchmarks serve as the starting point for setting standards and play a crucial role in offering incentives to industry to decarbonization (LKAB, 2020) (Respondent #6).

In accordance with the EU ETS, the European steel industry receives free allocation for five product benchmarks: coke, sinter, hot metal which is connected to the blast furnace process, EAF carbon steel, and EAF high alloy steel (EUROFER, 2022). These benchmarks play a key role in providing incentives to industry to decarbonize. The sinter plant is needed in a process that involves the conversion of iron ore fines into a mass, commonly known as a pellet (Jernkontoret, 2019). Conventionally, the sinter plant is situated on the same premises as the steelworks, i.e. BF-BOF. However, LKAB, a Swedish mining company and one of consortiums of HYBRIT has progressed to the next step in the value chain by including the pellet production rather than solely selling fines to steel producers (Respondent #6). LKAB is unique in its mining because its ore composition is mainly magnetite, which requires far less energy to transform it into steel than the hematite iron ore that is the main iron ore found elsewhere in the world (LKAB, 2020) (Respondent #6). Therefore, the emissions from LKAB's pelletizing process are significantly lower compared to other sinter work in Europe mainly due to the iron ore type. Despite the cleaner feature of LKAB's pellet compared to other European sinter process, they are currently not included in the sinter benchmark (ibid). The European steel industry's competitiveness in terms of reaching climate target is being negatively impacted by overcompensation, which is causing a decline in the competitiveness of those who have made significant investments in climate efficiency (ibid). The overcompensation could bring the investments in fossil-free steel, e.g. HYBRIT, a worse competitive position than the carbonintensive traditional steel industry (ibid). Nevertheless, as one interviewee expressed, there is hope as the Swedish government has been lobbying for the inclusion of the LKAB's pelletizing process in the sinter benchmark and would anticipate favorable results in the second period of EU ETS phase 4 starting in 2026.

Korea commenced the Korean ETS (K-ETS) in 2015, the first country in East Asia to do so, based on the laws on the Green Growth and the Act on Allocation and Trading of GHG Emission in Korea (GIR, 2022), and is now in the third phase of K-ETS (2021-2025). The steel industry in Korea has been so far receiving 100% free allowances ever since the K-ETS started and is expected to receive the 100% free allowances until the third phase. The specific details regarding the fourth phase of the K-ETS have yet to be laid out. However, in 2022, the ministry of Environment created an Emissions Trading Advancement Council to discuss ways to enhance the K-ETS to achieve the country's carbon neutrality and to address the Carbon Border Adjustment Mechanism (Influence Map, n.d.). The discussion pertains to decrease in free allocation, free allocation based on benchmark in the 4th phase (2026-2030) (ibid).

	EU ETS	K-ETS
Phase I	2005 – 2007 (3 years)	2015 – 2017 (3 years)
Phase II	2008 – 2012 (6 years)	2018 – 2020 (3 years)
Phase III	2013 – 2020 (8 years)	2021 – 2025 (5 years)
Phase IV	2021 – 2030 (10 years)	

Table 5-4 Timeframe of EU ETS & K-ETS

Source: Korea Exchange (2018)

5.2.4 Carbon Border Adjustment Mechanism (CBAM)

The gradual implementation of the CBAM is aligned with the phase-out of the allocation of free allowances under the EU ETS to facilitate the decarbonization of the EU industry (European Commission, n.d.-a). Both the EU ETS and CBAM are complemented one another, aiming at pricing GHG emissions embedded in the same sectors and goods. This is done via the use of specific allowances or certificates. Both systems have a regulatory nature and are justified by the need to mitigate GHG emissions, in line with the binding environmental target under the EU law, set out in Regulation (EU 2021/1119, which aims to reduce the EU GHG emissions (European Union, 2023). One of main differences between EU ETS and CBAM is that EU ETS applies to installation within the EU, whereas CBAM applies to certain goods imported into the customs territory of the EU (ibid). The CBAM will enter into force in its transitional phase as of October 2023 and targets carbon intensive and at most significant risk of carbon leakage goods. i.e. iron and steel, cement, aluminum, fertilizers, electricity and hydrogen during this phase (ibid).

It is true that there are some concerned voices. In the case of Korea, for instance, the steel industry is expected to be the most affected by the EU CBAM (KEEI, 2022). It is because the average share of steel exports to the EU during the 2018-2020 period was approximately 11% (MOTIE, 2023; KEEI, 2022.). Though the Korean government recently announced the Steel Industry Development Strategy with its investment plan to address potential challenges posed by the CBAM and to lay the foundation for the domestic steel industry, no definitive response has been observed from either the government or Korean steel producers. One respondent mentioned that the majority of the discussion in terms of how to respond CBAM so far has centered on the administrative procedures (Respondent #2). The respondent also mentioned that the government support tends to focus on SMEs rather than conglomerates. Since the EU announced the implementation of the CBAM in 2019, most studies have been conducted on the potential impact of the CBAM domestically and internationally (KEEI, 2022). While there has been some research on the economic impact of CBAM, however, there are relatively few studies on domestic measures on how to respond to CBAM (ibid). These few studies have focused on the SME perspective on how to respond to the CBAM, which shares the concern of one of the respondents about government support (ibid) as well as additional costs for the administrative process of CBAM.

5.3 Market formation

Bergek et al. categorizes three market phases for an emerging TIS market formation: nursing – in the very early stage, bridging – when volume increased and finally mature phase. For a TIS in a transformation phase, markets may not be fully formed (Bergek et al., 2008). In terms of the hydrogen-based steelmaking technology it was Sweden that first successfully piloting hydrogen-based steel production in 2018 and heading towards demonstration phase, thereby indicating the absence of current market for the large-scale hydrogen-based steelmaking technology. Therefore, the author perceives the current market for the hydrogen-based steel is nursing phase, or in other words, a niche market. To understand the sequence of the formation of markets, Bergek et al. suggested scrutinizing both actual market development and what drives market formation, for instance, the timing, size, and type of markets that have actually formed (ibid). Since the market size is still nursing phase, in order to investigate the market formation, demand signals, drivers of market formation are identified in the chapter.

The industrial sector is taking significant steps toward transition (IEA, 2022a). Many market research reports estimate the global green steel market was beyond USD 182 million in 2022 and is projected to grow rapidly ((Precedence Research, 2023; Vantage Market, 2023). The rising demand for low- to zero-emission steel is driven in part by climate targets set by consumers of steel producers such as construction and automotive, and others who seek to reduce their scope 3 emissions (IEA, 2022a). In addition, many automotive & construction companies, e.g., Lendlease, Skanska, Hyundai, Toyota, Ford, BMW, and others, announced their net zero target by 2050 at the latest. Given this widespread industry trend, the demand for low- emission steel will be high (ibid). This strong demand signal coming from steel offtakers has incited steel producers to react accordingly.

Moreover, there's a global initiative to drive global demand for green steel such as the SteelZero, the First Movers Coalition and etc. For example, SteelZero was launched by the Climate Group¹⁸, in collaboration with ResponsibleSteel, which is the first global initiative aimed at driving a demand signal for green steel (Green Steel World, 2023). In order to become a member, organizations need to make a public commitment to use, procure and specify 50% low-emission steel by 2030, with a long-term target of using 100% green steel (ibid). As of present, the current membership count of SteelZero amounts to 36 companies, including 4 Swedish companies namely Skanska UK, Vattenfall, ViaCon Group, Volvo Cars (SteelZero, n.d.). There are currently no Korean companies that are part of the SteelZero (ibid). In terms of policy to support this demand signal, NGOs and some other actors – but not necessarily coming from government side – discussing Green Public Procurement (GPP) as a tool to accelerate the market for green steel. In Sweden, or EU in broader term, is working on Eco Design Directive as a demand side measure (Respondent #7).

In case of Sweden, though the country has just passed the pilot stage and heads towards the demonstration stage meaning there is no current existing market for hydrogen-based steel, discussion on creating a market for decarbonized steel is actively on going. The HYBRIT project is aiming to scale up to a demonstration plant and to commission to make fossil free steel available to the market in 2026 (HYBRIT, n.d.). It has a plan to produce 1.2 Mt of crude steel per year, which represents 25% of Sweden's total production, and has the potential to reduce 14.3MtCO2eq of GHG emissions during the first ten years of operation (ibid). The production facility in Gällivare estimates to use 500 MW electrolyser capacity powered by fossil-free electricity. The project will enable to facilitate the replacement of two blast furnaces at SSAB with an electric arc furnace which would use the sponge iron as the feedstock to produce high-quality steel (HYBRIT, n.d.a). It is worth noting that the domestic market is gradually gaining recognition in Sweden. In 2021 summer, the first 100 tons of sponge iron from the pilot plant in Luleå were delivered to SSAB, which was then able to supply fossil-free steel to Volvo (HYBRIT, 2021b). SSAB and Volvo Group have signed an agreement to collaborate on

¹⁸ Climate Group - located in London, New York, New Delhi, Amsterdam and Beijing - is an international non-profit organization founded in 2003 working with governments and business to achieve net zero emissions by 2050 especially highemission sectors including energy, industry, transport and etc. ((Climate Group, n.d.).

research, development, serial production and commercialization of the world's first vehicles manufactured from fossil-free steel (SSAB, 2021). In 2022, SSAB made another progress in the development of fossil-free steelmaking and delivered around 500 tons of fossil-free steel to SSAB's customer. Volvo Construction Equipment (Volvo CE) became the first company in the world to deliver a construction machine using SSAB fossil-free steel to its customer, NCC¹⁹.

With respect to Korea, hydrogen-based steel production technology has not been piloted, however, in July 2022, POSCO signed a joint engineering business agreement with Primetals, a British plant construction company, to start designing the demonstration plant for hydrogenbased steel making HYREX (POSCO, 2022d). The reason POSCO can leapfrog to demonstration stage for hydrogen-based steel plant is that POSCO is currently operating a FINEX²⁰ facility that uses reducing gas containing 25% hydrogen that shares the basic principle with HYREX but uses full 100% hydrogen coming from renewable energy (ibid). HYREX follows the method of reducing iron ore by directly with hydrogen in a powdery state and can skip the pelletizing process and use relatively low-grade ore (ibid). POSCO aims to advance its H-DR technology by incrementally raising the hydrogen concentration in the two Pohang-based fluidized bed reactors to 1.5 million tons and 2 million tons per year, respectively (POSCO, 2021b). No specific companies have agreed to purchase hydrogen-based steel as an end-product, however, there has been active business-to-business interaction on hydrogen. POSCO and Hyundai Motor, for instance, signed a business agreement on hydrogen business cooperation. In principle, only one steelmaker's materials are used for one type of part, but different parts may contain materials supplied by different steel producers. Thus, Hyundai Motor's vehicles do not only use steel from Hyundai Steel, but it is known that steel from a wide variety of steelmakers is used in a single vehicle. Moreover, POSCO has its affiliated construction companies (Donga, 2021).

In February 2023, the Korean Ministry of Trade, Industry and Energy announced the strategy for the steel industry for low-carbon steelmaking transition and its plan to establish a fund worth 150 trillion won, equivalent to USD 116 million or EUR 105.3 million, to support steelmakers' transition to a low-emission production structure and to boost competitiveness (MOTIE, 2023; Yeonhap, 2023). This is almost the very first government's official support to the steel industry. In the last year 2022, MOTIE and the Korea Iron & Steel Association held the Iron Day celebration event and reached a business agreement to establish a steel ESG fund to support SMEs with ESG certifications. POSCO, Hyundai Steel and IBK Corporate bank that contributed a total of 150 billion won - 50 billion won, 20 billion won, and 80 billion won respectively (MOTIE, 2023; Chosun, 2022), but no government financial support for hydrogenbased iron & steelmaking technology was specified. In line with the ESG fund, this strategy set out 150 trillion won (equivalent to EUR 105.3 million) with the establishment of a private-public alliance involving major Korean steel companies, MOTIE, and Korea Iron & Steel Association as well as the development of a roadmap for low-carbon steel production roadmap in coming years. The strategy also set out its countermeasure for the Carbon Border Adjustment Mechanism, however, according to one interviewee, a specific roadmap that outlines specific plan will have been published in a couple of years since the announcement (Respondent #2).

¹⁹ NCC is a Swedish construction company that set climate neutral target by 2045 (NCC, 2022)

²⁰ FINEX is a method that uses iron ore and coal in powder form without the sintering and coking process which make raw materials into lumps that POSCO commercialized FINEX in 2007 (POSCO, 2022)

In case of Sweden, the HYBRIT was granted financial support of SEK 599 million (approx. EUR 52 million) by the Swedish Energy Agency through the Industriklivet initiative ²¹ (Vattenfall, 2019) and continuously has received investment for a research and technology development. The three shareholders of HYBRIT - SSAB, LKAB and Vattenfall – have also spent and committed to invest SEK 1.1 billion (ibid). The Swedish Energy Agency has granted HYBRIT an additional investment of SEK 250 million via the Industriklivet initiative, for another example, for building a rock cavern storage facility for fossil-free hydrogen gas on a pilot scale next to the current HYBRIT's pilot facility for direct reduction in Luleå (SSAB, 2021). In addition, the project received in total EUR 143 million for scale-up a complete value chain from mine to fossil free steel from the European Union Innovation Fund²² (HYBRIT, 2022b).

It is clear that the demand for a reduced emissions for steel products is growing, but steel producers' efforts need to be complemented by government and policy actors at an early stage. One of the main concerns about hydrogen-based steel products mentioned by some of interviewees was the additional premium cost and the willingness to bear the cost. Two potential policies that are under discussion as complementary measures for creating a market for green steel is Green Public Procurement (GPP). GPP is a policy instrument whereby public organizations seek to procure goods that have a reduced environmental impact throughout their lifecycle compared to similar goods that perform the same function (European Commission, 2016; Hasanbeigi & Bhadbhade, 2023). Government-funded construction and infrastructure project accounted for around 11% of total steel demand in Korea in 2019, and the government can use its purchasing power to stimulate demand for green steel products (Hasanbeigi & Bhadbhade, 2023).

	Korea	Sweden
Actors	Large, multinational firms	Large firms, start-up
Institutions	Inconsistence push & investment from government	Large amount of investment from government
Networks	Strong international learning network Emerging political networks Weak network	Strong domestic learning network Strong political network Organic cross-cutting network with all public-private sectors

Table 5-5 Comparison between Korea & Sweden

²¹ Industriklivet is a Swedish national initiative launched in 2017 with the aim of providing support for measures to mitigate Swedish process-related GHG emissions from the country's industrial sector as a step towards achieving the national netzero target. It offers funding for research, development and demonstration (Stockholm Environment Institute et al., 2022). Industriklivet is undertaking numerous projects in the Swedish industry sector including 25 projects in the steel industry (Industriklivet, n.d.).

²² The EU Innovation Fund has allocated a budget of approximately EUR 20 billion (subject to carbon prices) for financing the commercial demonstration of innovative low-carbon technologies over 2020-2030. The funding comes from the EU ETS and can support up to 60% of the additional CAPEX and OPEX of large-scale project. For smaller project, only CAPEX is eligible for coverage (Stockholm Environment Institute et al., 2022)

Knowledge	- 13 R&D and study reports	- 43 R&D and study reports
development and diffusion (R&D, investment)	- EUR 8.2 billion	- EUR 20-46 billion
Guidance of the search	Policy: Carbon pricing, ETS, CBAM	Policy: IDR, ETS
Market formation	Active B2B	Moderate B2B

Source: Author's own

6 Discussion

6.1 Comparison of the Korean and Swedish H-DR innovation systems

6.1.1 Discussion to RQ1: How has knowledge development and diffusion been supported in Korea VS Sweden?

The current level of knowledge development and diffusion in Korea for Green H-DR has been relatively low domestically in terms of the number of research papers, R&D, as well as the amount of investment for the R&D. Until the government designated POSCO's HYREX technology as a national R&D project in May 2023, it was mainly POSCO was responsible for paving path for Green H-DR without specific governmental support. However, the number of national projects which contain "H-DR" as a keyword was 2 and 3 research papers were derived from the national projects ((Kim & Sohn, 2022; Kwon et al., 2019; Park et al., 2020). These three research papers are of scientific and technological aspects which involve electrolysis, solid oxide fuel cell, porous carbon, poly (vinylidenefluoride), which are important parts of hydrogenreduction iron & steelmaking technology. However, research papers that overarchingly focus on H-DR technology diffusion in Korea that is publicly available were lacking. More comprehensive ones – i.e. steel industry decarbonization - were found from some grey literature from government research institutions, such as KEEI, KIET, Korea Institute of Energy Research (KIER), and a few reports from NGOs. With regards to the international knowledge sharing activities, on the other hand, POSCO has shown proactive approach to international cooperation - i.e. a joint study with Roy Hill & Hancock, the Australian mining company, hosting the international HYIS forum, and etc. - have been observed more frequently. For H-DR related R&D in Korea started with equivalent to 21 million EUR investment including recent government announcement.

One reason why the Korean domestic knowledge network has been weak, especially publicprivate, according to a respondent 1 from Korea, is that national projects are mostly driven by public-academia, which resulted in limited corporate engagement. However, when it comes to formulating carbon neutral strategy, large corporates participation is a must in order to identify demand for a certain technology and to determine the direction of technology and business, especially considering that large corporates play significant role in the Korean economy. In response to this need, corporates, including steel producers have expressed their interest to participation in national projects coordination, and their interest was received positively. However, respondent 1 expressed concern that this dialogue has not been consistent and may be subject to change depending on which political party take presidency (Korean respondent 1). Another reason, according to a respondent 2 from Korea, could be explained that the government's tendency to provide more support to small and medium-sized enterprises (SMEs) while remaining relatively indifferent towards conglomerates. The respondent also noted that, in turn, conglomerates are hesitant to share their data with the government (Korean Respondent 2). As climate target is our collective goal that needs concerted efforts, therefore, regardless of which political party take presidency, climate target should remain consistent so that corporates can really concentrate on their low-emission transition. Furthermore, corporates need to improve their data transparency, while also establish a comprehensive and detail roadmap.

In the case of Sweden, domestic knowledge development has been well established. Following the Paris Agreement, three main project phases were set: pre-feasibility study (2016-2017), feasibility study Pilot stage (2018-2024) and demonstration plant stage (2025-2035) (HYBRIT, 2017). In terms of R&D, a four-year (2017-2021) research program was launched with the support of the Swedish Energy Agency (HYBRIT, n.d.). It was observed during the interview that the three shareholders of the HYBRIT – SSAB, LKAB and Vattenfall - and the steel

association approached universities to collaborate in the HYBRIT project. It was mentioned by **three Swedish respondents** commonly that it is natural for corporates to approach academia for the purpose of such partnership in Sweden. The HYBRIT project was started with the EUR 51.5 million investment from SEA, and has resulted in 44 research papers. Domestic (or Scandinavian) network was strong, however, international knowledge network was not identified from HYBRIT. It is noteworthy initiative and groundbreaking milestone that Sweden has shown the world that green H-DRI is feasible technology. However, technology to decarbonize the steel industry is not only about one company or one country gaining competitive advantage over another but also about a collective effort to tackle climate change and decarbonize the world's most polluting industry. Therefore, this dialogue and cooperation should be expanded to more international dimension.

6.1.2 Discussion to RQ2: How has the guidance of search been supported in Korea and Sweden?

Despite the positive perception and recognition of the urgent need for the Green H-DRI, the technology transition in Korea has been relatively slow. One reason could be that each time there is a change in which party holds the presidency, the climate target and support also slightly changes based on the stance of the administration. For example, the current administration -President Yoon Seokyeol - announced the government's draft of the first National Carbon Neutrality and Green Growth Basic Plan (2023-2042) in 2022. It is the first sectoral GHG emissions reduction targets that the Yoon administration put forward. Under this plan, the GHG reduction target for industry sector has been set at 11.4% reduction by 2030, which is 3.1% less than the Moon administration's goal (14.5%) (Yeonhap, 2023). Respondent 1 mentioned that under the previous administration, carbon neutrality was more accelerated. Thus, according to the prevailing government's direction, POSCO announced its carbon neutral roadmap in 2021 and brought all energy intensive industry, all national projects together to commit to carbon neutral. Numerous corporates, including POSCO have put effort to prepare to this end. However, due to the government leadership change last year, this plan was postponed. The current administration has decreased the level of government pressure on both corporate entities and governmental investment with regards to carbon neutrality, in contrast to the previous administration. This raises another concern as the industry sector, including POSCO need to comply with the Nationally Determined Contributions (NDC) by 2030. Therefore, depending on the next government, steel producers could face significant pressure (Respondent 1). This clearly shows that inconsistency of government support for not only steel industry but for all corporates to reach climate targets presents not encouraging environment for corporates to accelerate their low-emission transition especially Green H-DR requires extensive support from all actors especially government.

In terms of regulation and policy, CBAM was the policy most frequently mentioned by interviewees as affecting the Korean steel industry. One interviewee expressed concern about CBAM saying that this is beyond corporates' capacity and requires government-to-government negotiation to respond CBAM. In February 2023, a roundtable discussion was held to discuss the steel industry's development strategy in response to CBAM (MOTIE, 2023). In the event, MOTIE, steel association, 7 steel producers including POSCO and Hyundai Steel agreed on creating a KRW 150 billion (EUR 107 million) fund to promote low-carbon steel production. CBAM also brought Korea and Japan together to cooperate with (MOTIE, 2023; Korean Herald, 2023). It is true that CBAM raised a concern of possibility of encouraging protectionism, however, it has finally brought government's attention to decarbonize the giant steel industry and support them that used to be absent. Watching Sweden leading in fossil-free steelmaking, and European Commission putting forward Carbon Border Adjustment Mechanism (CBAM) to prevent carbon leakage, it might have given a pressure to the Korean government to take

decarbonized transition to green steelmaking more seriously. (NEWSIS, 2022). What is more, it brought Korea and Japan together who have very sensitive diplomatic relation to cooperate.

In Sweden, three policy instruments were mentioned during the interviews: Industrial Emission Directive (IED), the Emission Trading Scheme (ETS) Directive and the Eco-Design directive. Starting with the IED, **respondent 1** expressed concern about the revised IED that it wants to set stricter emission limits without a thorough understanding of the differences between companies, sites and product types. Especially when making advanced products, the revised IED may not be advantageous in terms of energy efficiency and resource efficiency. The Swedish steel industry aims to produce advanced products of high quality as they are in the end better in terms of longer lifespan with less corrosion. **Another respondent (Respondent #2)** stated that IED and ETS are complementary. IED does not consider CO2 because it assumes that ETS takes it into account. But, due to free allocation and so on, it hasn't reduced CO2 emissions significantly. However, as the EU ETS is phasing out free allocation, legislators really need to make sure that these policy instruments do not double regulate the steel industry to shift to more advanced decarbonized technology.

6.1.3 Discussion to RQ3: How has the market formed for the green steel in Korea and Sweden?

Globally speaking, many networks are being formed to attract buyers in the green steel space, and promote market formation for green steel e.g. SteelZero, FirstMover Coalition, etc.. However, currently there are no Korean companies that are members of the international green steel coalition while those of Sweden have been identified. Along with this international initiative, the B2B contract is commonly recognized. For example, Korea, notably POSCO, signed MOUs with Hyundai Motors, while Swedish company SSAB has signed contracts with Volvo cars, Volvo Construction Equipment (Volvo CE), NCC, and etc.

In Korea, one respondent raised a concern about the additional premium cost associated with green steel as well as the lack of clarity related to consumer's willingness to purchase at an augmented cost. Some say that green steel will be 30% likely to be more expensive than conventional steel (Global Economic, 2023; Next Group & SFOC, 2022). However, some argue that as long as the green hydrogen can be produced with low price, green steel can remain similar cost range or cheaper (KEEI, 2022; Hydrogen Insight, 2023).

In Sweden, one respondent (Respondent #7) said that it is true that the bulk material industry, such as the steel industry, does not get incentives under the EU ETS to decarbonize, to change their process, when making advanced products in high quality. However, if there's a demand for green steel, steel industry is likely to answer that demand. Therefore, the respondent mentioned that the proposal for a new Ecodesign for sustainable product regulation (ESPR) is under discussion in the EU institutions as a demand side measure. Existing Ecodesign directive only applies to household energy using appliances, e.g. dishwasher, washing machine, etc., but the proposal tries to include intermediary products such as steel, to improve their circularity, energy performance and other environmental sustainability (European Commission, n.d.). This is to encourage consumers and businesses to make informed choices when purchasing products and improve transparency about life cycle impacts of products on the environment (ibid). If iron and steel are included, it will be part of product requirements applied to iron and steel products (Respondent #2). Though a clear instrument from EU or the government is not pronounced, a demand side measure for green steel in the market is currently on the discussion table among the EU institutions.

Several non-governmental organizations have proposed policy recommendations for green steel market through e.g., Green Public Procurement policies, but the Korean government have not yet publicly considered nor announced GPP as a tool to stimulate the market for green steel (Hasanbeigi & Bhadbhade, 2023). The same is true with Swedish government. One study on GPP at EU-level from the Stockholm Environment Institute says that the current implementation of GPP among EU member states remains below the target set by the EU Commission and the recent data is only available up to 2012 (Stockholm Environment Institute et al., 2023). And no discussion on GPP for green steel could be identified by the Swedish government nor respondents other than the Ecodesign Directive. GPP in Korea was first introduced in 2005, has grown fourfold from KRW 3.8 trillion to KRW 175.8 trillion by 2020, with civil engineering and building materials accounting for the largest share (46.1%) ((Next Group & SFOC, 2022). However, steel products are not included in the minimum green standard of the GPP product list (ibid). The public sector plays a key role in influencing and supporting the market formation for green steel as steel is one of the fundamental materials in all sectors, including the construction sector. Therefore, implementing policy support like GPP can provide reassurance to steel producers by sending a strong market signal for green steel.

Moreover, to diffuse this type of more sustainable and decarbonized technology, policy support is essential. The energy sector, for instance, to increase the deployment of renewable energy in 2000s, policy instrument such as the Feed-in-Tariff, significantly helped countries to increase the production of electricity from renewable sources in the initial stages. To be more specific, the first German Feed-in Tariff (EEG in German) was introduced in 2000 and contributed to the initial stage of increase in renewable energy deployment from 6.2% in 2000 up to 28% in 2014 (Future Policy, 2015). Although this example pertains to a different context, it demonstrates the necessity of policy instruments in promoting the adoption of new sustainable production technology in energy-intensive sectors like the steel industry. Therefore, the government involvement to facilitate the market for green steel especially in the early stage, is crucial.

6.2 Reflection on my study

The first and most important reason to choose this topic and delve into it was because I encountered much news about Sweden's success in piloting green H-DRI, whereas I came across many reports that the Korean steel industry emission reduction has been stagnated, and it is by far the most emitting industry in Korea and has been shielded from government's regulation such as free allocation. This made me ponder on why Korea is at a slow pace reducing GHG emissions contrary to its net zero target. However, this research has revealed that there are so many non-publicly spoken efforts within corporates to achieve decarbonization. And CBAM has accelerated in bringing the dispersed effort and consolidated even more where the effort used to be dispersed has now been brought attention from many different actors and network in Korea. Every country has its unique set of strength and weakness in order for technology diffuse, which can be influenced by factors such as political dynamics, trade advantages, geopolitical positioning, etc.. These factors can create advantages for some countries in specific areas, while others may face certain challenges. Based on the overall reflection, the following section presents the relevance of the study, selected methods, more specifically.

6.2.1 Methodological and analytical choices

The choice of a qualitative method with semi-structured interviews has allowed this research to provide an overarching insight into the challenges and trend of green H-DRI technology diffusion in the Korean and Swedish steel industries. The interview with steel producers, who

are directly related to green H-DR, allowed me to understand the challenge that the practitioners' concerns. Also, the interviews with NGOs helped me grasp some other aspects that are still significantly relevant but have not been actively brought to the surface such as the network in the steel sector, policies and perspectives of civil society and people of the area where steel plants are located. Using the TIS as a theoretical and analytical framework was appropriate as green H-DR is an emerging technology that succeeded in piloting stage that needs to be scaled up. As a theoretical framework, the TIS has laid the foundation for understanding the process of technological innovation and dissemination within a socio-technical system. It pointed to the interplay between technological, social, economic, and policy dimensions. Simultaneously, TIS as an analytical framework, it offers a various set of indicators to analyze each function on how the technology has diffused guiding the research to explore the role of key actors, institutions, networks. By examining knowledge development, policy interventions, and market interactions, I gained a comprehensive understanding of how both internal and external dynamics influence the adoption and scaling up of technology in distinct ways in each country's context. The TIS framework enabled me to reveal the complex interaction between external influences and internal capacities in driving the spread of H-DRI technology in the steel industry.

6.2.2 Legitimacy

The research questions that were formulated by using functions and indicators of the TIS was legitimate as green H-DR is an emerging technology that succeeded in pilot stage and needs to be scaled up. It is true that the TIS is sometimes criticized by its lack of attention to overall context, i.e. geographical issues, international relations, and marginal importance in politics and technology diffusion (Markard et al., 2015). Nevertheless, the TIS framework was the right choice to start as it provides more perspectives, i.e. economic and political aspects to analyze which allowed me to observe more than just technological perspective.

Further research could investigate the remaining functions of the TIS and integrate them into a comprehensive analysis. Moreover, as scaling up renewable electricity as well as green hydrogen production will play a pivotal role in green H-DRI, future research could investigate hydrogen governance, including trade dynamics, renewable electricity generation to produce hydrogen, its contribution to the electricity system and its implications/impact for the steel industry.

7 Conclusions

The steel industry is considered a hard to abate sector. However, HYBRIT project showed that steel industry is possible-to-abate using hydrogen though there's a mile away to scale up. This paper compared steel industry in Korea and Sweden in terms of green H-DR technology transition analyzing three functions in the technological innovation systems: 1) knowledge development, 2) guidance of search and 3) market formation. Figure 7-1 depicts a summary of the domestic and international collaboration levels observed in Sweden and Korea. The study reveals that Sweden exhibits robust domestic collaboration but limited international cooperation. Conversely, Korea demonstrates a strong international cooperation and a weak domestic cooperation.

Sweden identified relevant actors and formed a strong domestic network initiated collaborated R&D with universities, research institutes etc. Sweden already published the steel industry roadmap to decarbonize in 2019. There has been EU-level policies that impacted the steel industry decarbonization and Sweden itself is known for its high carbon price. There hasn't been explicit support to foster demand and formation of market for green steel, however, there has been active B2B contract with green steel companies and vehicles and construction companies. The main message from the Swedish case is that a successful TIS needs stable institutional conditions (Swedish Energy Agency & the EU Commission), where actors are supported to obtain knowledge and skills from experimenting with the technology, form strong networks (universities, research institutes, corporates, etc.) that have an impact on learning process and policy making with consistent investment.

Korea has a strong international position, and this has been observed through 1) POSCO's international network for its supply chain with Australia and Oman, and 2) POSCO creating an international platform for knowledge development networking, e.g., HYIS forum. However, domestic knowledge network is weak green H-DR technology that POSCO was trying to promote did not gain much attention until recently. ETS is the most relevant policy that affects the steel industry in Korea, however, ETS has not been successfully reduce the GHG from steel industry. Instead, CBAM really gathered all actors in industry sector in Korea to support the steel industry decarbonization to maintain the competitiveness. The international market appears to be more attractive since there is no explicit domestic demand, other than one Korean vehicle company, nor market for green steel. As cost is the main competitiveness of Korean steel, therefore it is the matter of who would buy increased cost of green steel. For the Korean steel industry has suffering from inconsistent policies, investment and commitment depending on the presidency. In foreign markets, CBAM was once concerning but it has accelerated domestic actors to engage especially government's support. International knowledge exchange appears to be important for Korean R&D in order to stay in a leading position worldwide with respect to research. International markets appear to be just as attractive for the Swedish steel companies, but B2B contract is already made in the domestic market.



Figure 7-1 Comparison H-DR TIS between Sweden & Korea

Source: Author's own

7.1 Recommendations for future research

Based on the findings through this research, this section will delineate some recommendations for future research.

Strengthen alliances

As CBAM and Inflation Reduction Act (IRA) from the US are expected to bring about reducing greenhouse gas in energy-intensive industry, however, there is a potential risk of protectionism. Therefore, it's a worth consider for Korea to cooperate with neighboring countries and create and strengthen alliances. The recent initiatives in meeting with Korea and Japan governmental higher-ups as well as active trade with Australia shows positive move on this front. Thus, establishing an Asia-Oceania steel alliance could facilitate steel industry decarbonization globally.

Enhance international cooperation

Building upon the previous recommendation, beyond compartmentalizing efforts by region, the global steel industry should foster cooperation to collectively address environmental and climate challenges together. A unified approach across the industry will be more effective in achieving meaningful and sustainable decarbonization goals. The green steel networks and the HYIS forum are good examples.

Policy support, e.g., Green Public Procurement

The research found out that neither Korea nor Sweden have not yet actively considered GPP for green steel promotion in the market. Recognizing the significant influence do not actively consider GPP for green steel. As the public sector has a strong purchasing power to the market, countries need to revisit their GPP policy and foster the demand for green steel in the market.

 $Continuous \ investment \ R\&D-consistent \ with \ university \ and \ private \ and \ pulic$

The research found out one concern in Korea was that depending on which party taking the presidency, the pressure on meeting climate targets and the amount of investment have been inconsistent. Another Korean respondent in the research mentioned that government's support tends to be biased towards boosting SMEs growth, giving less attention to conglomerates. However, although green steel transition is dominantly led by conglomerates, consistent government engagement and support is a must to reach industrial and national climate target. Therefore, consistency in governmental support and investment is a must regardless of the political party in power.

Conduct Interdisciplinary Research

Steels are used everywhere ranging from, white home appliances to building materials, ships, wind turbines. Steel, from the production process to the finished products, has so much impact on society and people's daily life. For example, conflict with local communities over the construction of wind farms and the rights of the indigenous Sami people has caused controversy. Therefore, green steel transition needs a more interdisciplinary approach that goes beyond technological aspects and integrate socio-economic, socio-political and various other dimensions.

Bibliography

- Agora Industry, & Wuppertal Institute. (2023). 15 insights on the global steel transformation. https://static.agora-energiewende.de/fileadmin/Projekte/2021/2021-06_IND_INT_GlobalSteel/A-EW_298_GlobalSteel_Insights_WEB.pdf
- Åhman, M., & Nilsson, L. J. (2015). Decarbonising industry in the EU climate, trade and industrial policy strategies. *Palgrave Macmillan*, 92–114.
- Åhman, M., Olsson, O., Vogl, V., Nyqvist, B., Maltais, A., Nilsson, L. J., Halding, K., Skaneberg, K., & Nilsson, M. (2018). Hydrogen steelmaking for a low-carbon economy. *LTH, Lunds Universitet*. https://portal.research.lu.se/en/publications/hydrogen-steelmakingfor-a-low-carbon-economy-a-joint-lu-sei-work

Anderson, J. (2022). EMISSIONS REDUCTIONS IN THE STEEL INDUSTRY.

- Bergek, A., Hekkert, M., & Jacobsson, S. (2008). Functions in innovation systems: A framework for analysing energy system dynamizes and identifying goals for system building activities by entrepreneurs and policy makers. www.chalmers.se/tme/EN/centers/ride
- Bergek, A., & Jacobsson, S. (2003). The emergence of a growth industry: A comparative analysis of the German, Dutch and Swedish wind turbine industries. In J. S. Metcalfe & U. Cantner (Eds.), *Change, Transformation and Development* (pp. 197–227). Physica-Verlag HD. https://doi.org/10.1007/978-3-7908-2720-0_12
- Bergek, A., Jacobsson, S., Carlsson, B., Lindmark, S., & Rickne, A. (2008). Analyzing the functional dynamics of technological innovation systems: A scheme of analysis. *Research Policy*, 37(3), 407–429. https://doi.org/10.1016/j.respol.2007.12.003
- Bulkeley, H., Stripple, J., Nilsson, L. J., van Veelen, B., Kalfagiann, A., Bauer, F., & Sluisveld,M. A. E. van. (2022). *Decarbonising economies*. Cambridge University Press.
- CAN Europe. (2022). CAN Europe's transformation pathway recommendations for the steel industry. https://caneurope.org/content/uploads/2022/06/CAN-Europe_2022_Recommendations_Steel_Transforming-the-sector.pdf
- Carlsson, B., & Stankiewicz, R. (1991). On the nature, function and composition of technological systems. *Journal of Evolutionary Economics*, 1(2), 93–118. https://doi.org/10.1007/BF01224915

- Ćetković, S., & Stockburger, J. (2023). Hydrogen Strategy of Sweden: Unpacking the Multiple Drivers and Barriers to Hydrogen Development. *RIFS Potsdam*.
- Chosun. (2023, April 13). [단독] 오만에 8조원 그린수소 플랜트 사업 수주 유력.

 조선일보.
 https://www.chosun.com/national/transport

environment/2023/04/14/SS2BE2RGINB6HHBBZNXEBVI24I/

- Creswell, J. W. (2018). Research Design: Qualitative, Quantitative, and Mixed Methods Approaches.
- Di ez, M. A., Alvarez, R., & Barriocanal, C. (2002). Coal for metallurgical coke production: Predictions of coke quality and future requirements for cokemaking. *International Journal* of Coal Geology, 50(1–4), 389–412. https://doi.org/10.1016/S0166-5162(02)00123-4
- Donga. (2021, February 20). 수소가 뭐길래...현대제철 가진 현대차가 포스코 찾아간 이유 [김도형 기자의 휴일차(車)담]. 동아일보. https://www.donga.com/news/article/all/20210220/105531620/1
- EDF. (2016). *Electricity Market*. https://cermics.enpc.fr/~delmas/Enseig/levy-delatourmarket.pdf
- Edquist, C. (2006). Systems of Innovation: Perspectives and Challenges. Oxford University Press. https://doi.org/10.1093/oxfordhb/9780199286805.003.0007
- ENGIE. (n.d.). Carbon Contracts For Difference To Boost Industrial Decarbonization. ENGIE Impact. Retrieved September 28, 2023, from https://www.engieimpact.com/insights/carboncontracts-for-difference
- EUROFER. (2021). Position paper on Contracts for Difference.
- EUROFER. (2022). Commission proposal on ETS: Explaining the benchmark issue.
- European Commission. (n.d.-a). Carbon Border Adjustment Mechanism. Retrieved May 31, 2023, from https://taxation-customs.ec.europa.eu/carbon-border-adjustmentmechanism_en
- European Commission. (n.d.-b). Industrial Emissions Directive. Retrieved June 10, 2023, from https://environment.ec.europa.eu/topics/industrial-emissions-and-safety/industrial-emissions-directive_en

- European Commission. (2014). Country Report -Sweden. https://energy.ec.europa.eu/system/files/2014-10/2014_countryreports_sweden_0.pdf
- European Commission. (2015). EU ETS Handbook. http://ec.europa.eu/clima/policies/ets/index_en.htm
- European Commission. (2016). Buying green!: A handbook on green public procurement: 3rd edition. Publications Office. https://data.europa.eu/doi/10.2779/246106
- European Commission. (2021). Update of benchmark values for the years 2021-2025 of phase 4 of the EU ETS. https://climate.ec.europa.eu/system/files/2021-10/policy_ets_allowances_bm_curve_factsheets_en.pdf

European Commission. (2023). Questions and Answers: Carbon Border Adjustment Mechanism.

- Directive 2010/75/EU of the European Parliament and of the Council of 24 November 2010 on industrial emissions (integrated pollution prevention and control) (Recast) (Text with EEA relevance), (2011). http://data.europa.eu/eli/dir/2010/75/2011-01-06/eng
- European Union. (2023). Official Journal of the European Union. https://eur-lex.europa.eu/legalcontent/EN/TXT/PDF/?uri=OJ:L:2023:130:FULL
- Eurostat. (2023). *Glossary:Emissions trading system (ETS)*. https://ec.europa.eu/eurostat/statistics-explained/index.php?title=Glossary:Emissions_trading_system_(ETS)
- Fossil Free Sweden. (2021). Strategy for fossil free competitiveness—Hydrogen. https://fossilfrittsverige.se/wp-content/uploads/2021/01/Hydrogen_strategy_for-_fossil_free_competitiveness_ENG.pdf
- Freeman, C. (1987). Technology, Policy, and Economic Performance: Lessons from Japan. Pinter Publishers.
- Gerres, T., Lehne, J., Mete, G., Schenk, S., & Swalec, C. (n.d.). Green steel production: How G7 countries can help change the global landscape.
- Gerres, T., & Linares, P. (2022). Carbon Contracts for Differences (CCfDs) in a European context.GIR.(2022).NationalStatistics—GHGinformation.https://www.gir.go.kr/home/board/read.do?pagerOffset=0&maxPageItems=10&ma

xIndexPages=10&searchKey=&searchValue=&menuId=36&boardId=58&boardMas terId=2&boardCategoryId=

Global Economic. (2023, May 26). *그린스틸 생산 비용 30% 급증...전 세계 철강업체들의 고민*. 글로벌이코노믹. https://www.genews.com/article/Industry/2023/05/202305261528343928e8b8a793f7_1

- Global Energy Monitor. (2022a). POSCO Gwangyang steel plant. https://www.gem.wiki/POSCO_Gwangyang_steel_plant
- Global Energy Monitor. (2022b). POSCO Pohang steel plant. https://www.gem.wiki/POSCO_Pohang_steel_plant
- Green Hydrogen Organisation. (n.d.). Sweden. Green Hydrogen Organisation. Retrieved September 7, 2023, from http://gh2.org/countries/sweden
- Green Steel World. (2023, March 14). SteelZero: A strong demand signal accelerating the net zero transition. *GreenSteelWorld.Com*. https://greensteelworld.com/steelzero-a-strong-demand-signal-accelerating-the-net-zero-transition
- Greenpost Korea. (2022). POSCO receives free allowances, "Vulnerable as an exporter" [포스코, 탄소배출권 무상할당 받아..."수출기업으로서 취약"]. Greenpost Korea. http://www.greenpostkorea.co.kr/news/articleView.html?idxno=202388
- Griffin, P. W., & Hammond, G. P. (2021). The prospects for 'green steel' making in a net-zero economy: A UK perspective. *Global Transitions*, 3, 72–86. https://doi.org/10.1016/j.glt.2021.03.001
- H2GS. (2022, August 19). H2 Green Steel and BMW Group sign final agreement on delivery of CO2reduced steel. H2 Green Steel. https://www.h2greensteel.com/latestnews/h2-green-steeland-bmw-group-sign-final-agreement-on-delivery-of-co2-reduced-steel

Hankyung. (2022, March 29). 포스코, "로이힐 파트너" 행콕과 수소환원제철 원료 생산 나선다. 한국경제.

https://www.hankyung.com/economy/article/202203298541i

Hankyung. (2023, June 16). Korea Hydrogen Alliance invests 8 trillion KRW to build green hydrogen plant in Oman [韓수소연합, 8兆 투자...오만에 그린 수소 플랜트 건설]. 한국경제. https://www.hankyung.com/economy/article/2023061652991

- Hasanbeigi, A., & Bhadbhade, N. (2023). Green Public Procurement of Steel in India, Japan, and South Korea.
- Hekkert, M. P., Suurs, R. A. A., Negro, S. O., Kuhlmann, S., & Smits, R. E. H. M. (2007).
 Functions of innovation systems: A new approach for analysing technological change. *Technological Forecasting and Social Change*, 74(4), 413–432.
 https://doi.org/10.1016/j.techfore.2006.03.002
- Henrekson, M., Sandström, C., & Terjesen, S. (2021, April 12). Closing remarks in the HYBRIT debate: Current policies hinder rather than help sustainable development [Slutreplik i Hybritdebatten: Nuvarande politik stjälper snarare än hjälper hållbar utveckling]. Dagens industri. https://www.di.se/debatt/slutreplik-i-hybritdebatten-nuvarande-politik-stjalpersnarare-an-hjalper-hallbar-utveckling/
- Hong, S., & Kim, S. (2022). Evaluating the sustainability of the hydrogen economy using multi-criteria decision-making analysis in Korea. https://nextgroup.or.kr/nextkor/upfile/P220419163612.pdf
- HYBRIT. (n.d.). Research within HYBRIT. Hybrit. Retrieved March 10, 2023, from https://www.hybritdevelopment.se/en/research-project-1/
- HYBRIT. (2021a). A fossil-free future. *Hybrit*. https://www.hybritdevelopment.se/en/a-fossil-free-future/
- HYBRIT. (2021b, April 7). HYBRIT: SSAB, LKAB and Vattenfall building unique pilot project in Luleå for large-scale hydrogen storage investing a quarter of a billion Swedish kronor. *Hybrit.* https://www.hybritdevelopment.se/en/april-7-2021-hybrit-ssab-lkab-and-vattenfall-building-unique-pilot-project-in-lulea-for-large-scale-hydrogen-storage-investing-a-quarter-of-a-billion-swedish-kronor/
- HYBRIT. (2022a, February 18). HYBRIT: A unique hydrogen storage facility in Luleå is taking shape. *Hybrit*. https://www.hybritdevelopment.se/en/hybrit-a-unique-hydrogenstorage-facility-in-lulea-is-taking-shape/
- HYBRIT. (2022b, April 1). HYBRIT receives support from the EU Innovation Fund. *Hybrit*. https://www.hybritdevelopment.se/en/hybrit-receives-support-from-the-euinnovation-fund/

- HYIS. (2022). H2 Iron & Steel Making Forum 2022. H2 Iron & Steel Making Forum 2022. https://www.h2ironsteelforum.com/
- IEA. (2019). Energy Policies of IEA Countries—Sweden 2019 Review.
- IEA. (2020a). Iron and Steel Technology Roadmap—Towards more sustainable steelmaking.
- IEA. (2021). Korea Electricity Security Review. https://www.iea.org/reports/korea-electricity-security-review
- IEA. (2022a). Achieving Net Zero Heavy Industry Sectors in G7 Members. OECD. https://doi.org/10.1787/f25c9648-en
- IEA. (2023). Renewable Hydrogen from Oman.
- IEA. (2020b). ETS in industry Implementing Effective Emissions Trading Systems Analysis. IEA. https://www.iea.org/reports/implementing-effective-emissions-trading-systems/etsin-industry
- IEA. (2022b). Sweden-Countries & Regions. IEA. https://www.iea.org/countries/sweden
- IEA, & KEEI. (2022). Reforming Korea's Electricity Market for Net Zero. OECD. https://doi.org/10.1787/47d4e925-en
- InfluenceMap. (n.d.). *Korea Emissions Trading Scheme (K-ETS)*. Retrieved May 31, 2023, from https://korea.influencemap.org/policy/-3600bccb14d8b5d92cd3cfb557a8be1e-1830
- IPCC. (2022). Climate Change 2022 Mitigation of Climate Change.
- Jernkontoret. (2018). Klimatfärdplan: För en fossilfri och konkurrenskraftig stålindustri i Sverige. Jernkontoret.
- Jernkontoret. (2019). Climate issues and carbon emissions trading. Climate Issues and Carbon Emissions Trading - Jernkontoret. https://www.jernkontoret.se/en/energy-environment/climate-issues-and-carbon-emissions-trading/
- Jernkontoret. (2021). Industrial Emissions Directive and BAT Conclusions for the steel industry. Industrial Emissions and BAT Conclusions - Jernkontoret. https://www.jernkontoret.se/en/energy--environment/industrial-emissions-directiveand-bat-conclusions/

Jernkontoret, & Sweden Institute for Standard. (2023). Decarbonization standards in the steel sector – why, what and who—YouTube.

https://www.youtube.com/live/RHU93pZoiMo?feature=share

- Kang, B. (2022). Scenario analysis of iron and steel production process for carbon neutrality [탄소중립을 위한 철강 생산공정 전환 시나리오 분석 연구]. Korea Energy Economics Institute.
- KEEI. (2022). A study on the possibility of improving climate and trade policies to respond to carbon border adjustment. Korea Energy Economics Institute.
- Kim, W., & Sohn, I. (2022). Critical challenges facing low carbon steelmaking technology using hydrogen direct reduced iron. *Joule*, 6(10), 2228–2232. https://doi.org/10.1016/j.joule.2022.08.010
- Korea Energy Economics Institute. (n.d.). *National Energy Plan*. Retrieved March 8, 2023, from http://www.keei.re.kr/main.nsf/index_en.html?open&p=%2Fweb_keei%2Fen_Issue s01.nsf%2FByview04V%2F0743694E09C2AD4E49256E2900483FB0&s=%3FOpen Document
- Korea Exchange. (2018). ETS Economic Mechanism [배출권거래제의 경제적 메커니즘]. https://pdf.krx.co.kr/ebook/access/ecatalog_pt.jsp?Dir=42&catimage=&cpage=&ca llmode=
- Korea Institute for Industrial Economics & Trade. (2022). 철강산업의 탄소중립 추진 전략과 정책과제 [Carbon neutrality strategies and policy challenges for the steel industry].
- Korean Government. (2019). Hydrogen Economy Roadmap [수소경제 활성화 로드맵].
- KPX. (2020). *Electricity Market Statistics [전력시장통계*]. https://epsis.kpx.or.kr/epsisnew/selectEkifBoardList.do?menuId=080401&boardId= 040100
- Kushnir, D., Hansen, T., Vogl, V., & Åhman, M. (2019). Adopting hydrogen direct reduction for the Swedish steel industry: A technological innovation system (TIS) study. https://doi.org/10.1016/j.jclepro.2019.118185
- Kwon, Y., Yoo, J. Y., Jang, Y., & Bae, J. (2019). Long-term durability of La0.75Sr0.25Cr0.5Mn0.5O8-3 as a fuel electrode of solid oxide electrolysis cells for co-
electrolysis. Journal of CO2 Utilization, 31, 192–197. https://doi.org/10.1016/j.jcou.2019.03.004

Kyongbuk Ilbo. (2023, August 7). POSCO starts construction of H-DR in line with the global carbon neutrality trend [철강업계 글로벌 "탄소 중립" 트렌드 맞춰 포스코그룹, 수소환원제철소 건설 본격화]. 경북일보 - 굿데이 굿뉴스.

http://www.kyongbuk.co.kr/news/articleView.html?idxno=2138974

- LKAB. (2020, March 5). Leading the way to decarbonisation. LKAB. https://lkab.com/en/news/leading-the-way-to-decarbonisation/
- LKAB. (2022). Annual and sustainability report 2021. https://arsredovisning.lkab.com/wpcontent/uploads/2022/04/LKAB-Annual-And-Sustainability-Report-2021.pdf#page=37
- MacDonald, F. (2015, September 25). Sweden Sets Its Sights on Becoming The World's First Fossil Fuel-Free Nation. ScienceAlert. https://www.sciencealert.com/sweden-sets-its-sights-onbecoming-the-world-s-first-fossil-fuel-free-nation
- MaintWorld. (2019). SEK 200 million invested in pilot plant for storage of fossil-free hydrogen in Luleå. https://www.maintworld.com/R-D/SEK-200-million-invested-in-pilot-plant-forstorage-of-fossil-free-hydrogen-in-Lule
- McVeigh, K. (2022, August 10). 'We borrow our lands from our children': Sami say they are paying for Sweden going green. *The Guardian*. https://www.theguardian.com/global-development/2022/aug/10/indigenous-sami-reindeer-herders-sweden-green-transition
- Merriam, S. B. (1998). *Qualitative research and case study applications in education* (2nd ed). Jossey-Bass Publishers.
- Ministry of Climate and Enterprise. (2021, March 11). Sweden's climate policy framework [Text]. Regeringskansliet; Regeringen och Regeringskansliet. https://www.government.se/articles/2021/03/swedens-climate-policy-framework/ Ministry of Environment. (2023). 국내기업, 오만에 '그린수소 생산기지' 건설...세계

최대 규모. https://www.korea.kr/news/policyNewsView.do?newsId=148916681

- Ministry of Foreign Affairs. (n.d.). *Energy*. Retrieved September 5, 2023, from https://www.mofa.go.kr/eng/wpge/m_5657/contents.do
- Ministry of Government Legislation. (2021). *Electric Utility Act.* https://www.law.go.kr/LSW/lsInfoP.do?lsiSeq=232909&urlMode=engLsInfoR&vie wCls=engLsInfoR#EJ25:0
- MOTIE. (2023a). The 10th Basic Plan for Long-term Electricity Supply and Demand[제10차 전력수급기본계획].
- MOTIE. (2023b). The Korean steel industry development strategy on low-carbon steelmaking transition [저탄소 철강생산 전환을 위한 철강산업 발전전략].
- MOTIE. (2021). The 1st Hydrogen Economy Implementation Basic Plan [1차 수소경제 이행 기본계획].

https://www.motie.go.kr/motie/ne/announce2/bbs/bbsView.do?bbs_seq_n=67130 &bbs_cd_n=6

- MOTIE. (2023c). Decarbonizing steel production [철강생산 저탄소화...1500 억 민간펀드 만든다]. https://www.korea.kr/news/policyNewsView.do?newsId=148911821
- MSIT. (2023). Korea's Top 100 Carbon Neutral Technologies [한국형 탄소중립 100대 핵심기술 확정].

https://www.msit.go.kr/bbs/view.do?sCode=user&mId=239&mPid=113&bbsSeqN o=94&nttSeqNo=3183074

- Muslemani, H., Liang, X., Kaesehage, K., Ascui, F., & Wilson, J. (2021). Opportunities and challenges for decarbonizing steel production by creating markets for 'green steel' products. *Journal of Cleaner Production*, 315, 128127. https://doi.org/10.1016/j.jclepro.2021.128127
- Newsway. (2020, December 13). POSCO to produce 5 million tonnes of hydrogen by 2050... Targeting \$30 trillion in sales [포스코, 2050년 수소 500만톤 생산체제 구축…매출 30조 목표]. 뉴스웨이.

https://www.newsway.co.kr/news/view?ud=2020121315370293218

Next Group, & SFOC. (2022). Revisiting Korean Green Public Procurement Policies to Promote Green Steel Demand.

NKIS. (n.d.). National Knowledge Information System. Retrieved July 27, 2023, from https://www.nkis.re.kr/eng/introduce.do#

NTIS. (n.d.). About NTIS. Retrieved June 4, 2023, from https://www.ntis.go.kr/ThAbout.do

- Park, H.-Y., Lee, C. H., Cho, D.-W., Lee, C.-H., & Park, J.-H. (2020). Synthesis of porous carbon derived from poly(vinylidenefluoride) and its adsorption characteristics for CO2 and CH4. *Microporous and Mesoporous Materials*, 299, 110121. https://doi.org/10.1016/j.micromeso.2020.110121
- Pei, M., Petajaniemi, M., Regnell, A., & Wijk, O. (2020). Toward a Fossil Free Future with HYBRIT: Development of Iron and Steelmaking Technology in Sweden and Finland.
- POSCO. (2008). Sustainability report 2008. http://www.posco.com/homepage/sustainability/report/
- POSCO. (2010). Carbon Report 2010 Toward Four Greats. www.posco.co.kr
- POSCO. (2017a). POSCO report 2016.
- POSCO. (2018). POSCO's 50 Years of History.
- POSCO. (2020). POSCO's Dialogue for Climate Action 2020.
- POSCO. (2021a). POSCO Corporate Citizenship Report. https://www.posco.co.kr/homepage/docs/eng7/jsp/irinfo/irdata/s91b6000030l.jsp
- POSCO. (2015). *How POSCO Became POSCO the Great*. https://newsroom.posco.com/en/how-posco-became-posco-the-great/
- POSCO. (2017b). Discover the Technology that is Making Steel Production More Sustainable. https://newsroom.posco.com/en/discover-the-tech-making-steel-more-sustainable-finex/
- POSCO. (2019). Blast Furnace Anatomy #1 To the Heart of Steelworks Operation. https://newsroom.posco.com/en/blast-furnace-anatomy-1/
- POSCO. (2021b). The future of the hydrogen industry and POSCO's challenge. https://newsroom.posco.com/en/the-future-of-the-hydrogen-industry-and-poscoschallenge/

- POSCO. (2022a). [Great Conversion to Low-carbon Eco-friendly Steelmaking Process] ① HyREX. https://newsroom.posco.com/en/special-project-for-great-conversion-to-lowcarbon-eco-friendly-steelmaking-process-%e2%91%a0-hyrex/
- POSCO. (2022b). POSCO Examines Feasibility of Production of Low Carbon Steel Materials with Australian resource development company. https://newsroom.posco.com/en/poscoexamines-feasibility-of-production-of-low-carbon-steel-materials-with-australianresource-development-company/
- POSCO. (2022c). POSCO starts to design the HyREX demonstration plant. https://newsroom.posco.com/en/posco-starts-to-design-the-hyrex-demonstrationplant/
- POSCO. (2022d). POSCO starts to design the HyREX demonstration plant. https://newsroom.posco.com/en/posco-starts-to-design-the-hyrex-demonstrationplant/
- POSCO Holdings. (n.d.). *History*—POSCO HOLDINGS. Retrieved August 14, 2023, from https://www.posco-inc.com:4453
- POSCO International. (n.d.). POSCO INTERNATIONAL CORPORATION. Retrieved September 6, 2023, from http://www.poscointl.com/
- Precedence Research. (2023). Green Steel Market Size To Hit USD 6,24,414.85 Mn By 2032. https://www.precedenceresearch.com/green-steel-market
- Project Syndicate. (2022, December 13). *Climate Before Trade* | *by Dani Rodrik*. Project Syndicate. https://www.project-syndicate.org/commentary/fighting-climate-change-priorityover-wto-trade-rules-by-dani-rodrik-2022-12
- Regeringskansliet, R. och. (2014, November 24). Ministry of Climate and Enterprise [Text]. Regeringskansliet; Regeringen och Regeringskansliet. https://www.government.se/government-of-sweden/ministry-of-climate-andenterprise/
- Richstein, J. C. (2017). Project-Based Carbon Contracts: A Way to Finance Innovative Low-Carbon Investments. *SSRN Electronic Journal*. https://doi.org/10.2139/ssrn.3109302

- Richstein, J. C., & Neuhoff, K. (2022). Carbon contracts-for-difference: How to de-risk innovative investments for a low-carbon industry? *iScience*, 25(8), 104700. https://doi.org/10.1016/j.isci.2022.104700
- RIST. (n.d.). *History of RIST*. Retrieved October 26, 2023, from https://www.rist.re.kr/eng/class/intro/intro_02.jsp?floc=2
- Sekiguchi, N. (2017). Trade specialisation patterns in major steelmaking economies: The role of advanced economies and the implications for rapid growth in emerging market and developing economies in the global steel market. *Mineral Economics*, 30(3), 207–227. https://doi.org/10.1007/s13563-017-0110-2
- Sparke, L. (2022, March 31). POSCO, Hancock partner for low-carbon steel materials production. Roy Hill. https://www.royhill.com.au/posco-hancock-partner-for-low-carbon-steelmaterials-production/
- SSAB. (n.d.). *Time line for fossil-free steel production*. SSAB. Retrieved August 17, 2023, from https://www.ssab.com/en/company/sustainability/first-in-fossil-free-steel/timeline
- SSAB. (2017). *SSAB aims to be fossil-free by 2045*. SSAB. https://www.ssab.com/en/news/2017/10/ssab-aims-to-be-fossilfree-by-2045
- SSAB. (2021). Volvo Group and SSAB to collaborate on the world's first vehicles of fossil-free steel. SSAB. https://www.ssab.com/en/news/2021/04/volvo-group-and-ssab-to-collaborate-onthe-worlds-first-vehicles-of-fossilfree-steel
- Strbac, G., & Wolak, F. A. (2017). Electricity Market Design and Renewables Integration in Developing Countries.
- Swedish Energy Agency. (2021a). Industry- current situation and conditions for transition [Industrin nuläge och förutsättningar för omställning]. https://www.bing.com/ck/a?!&&p=90880d47877febc9JmltdHM9MTY5MjQ4OTY wMCZpZ3VpZD0zYzgwMTI2MS0zMDAwLTY0ODUtMmY0MC0wMTM3MzE4Z TY1OTEmaW5zaWQ9NTE3NQ&ptn=3&hsh=3&fclid=3c801261-3000-6485-2f40-0137318e6591&psq=Industrin+%e2%80%93+nul%c3%a4ge+och+f%c3%b6ruts%c 3%a4ttningar+f%c3%b6r+omst%c3%a4llning%3a+En+nul%c3%a4gesanalys+inom +Industriklivet.&u=a1aHR0cHM6Ly9lbmVyZ2lteW5kaWdoZXRlbi5hLXcybS5zZS9

Gb2xkZXJDb250ZW50cy5tdmMvRG93bmxvYWQ_UmVzb3VyY2VJZD0yMDI4N zQ&ntb=1

Swedish Energy Agency. (2022). Energy in Sweden 2022 – An overview.

- Swedish Energy Agency. (2021b). Proposal for a national strategy for fossil-free hydrogen [Förslag till nationell strategi för fossilfri vätgas]. https://www.energimyndigheten.se/nyhetsarkiv/2021/forslag-till-nationell-strategifor-fossilfri-vatgas/
- Ulku, H. (2004). R&D, Innovation, and Economic Growth: An Empirical Analysis. IMF Working Paper. https://ssrn.com/abstract=879010
- Van de Ven, & Garud. (1987). A framework for understanding the emergence of new industries. https://www.researchgate.net/publication/244955957_A_framework_for_understand ing_the_emergence_of_new_industries
- Vantage Market. (2023). Green Steel Market Size USD 120004.8 Million by 2030. Vantage Market Research. https://www.vantagemarketresearch.com
- Vasseur, V., Kamp, L. M., & Negro, S. O. (2013). A comparative analysis of Photovoltaic Technological Innovation Systems including international dimensions: The cases of Japan and The Netherlands. *Journal of Cleaner Production*, 48, 200–210. https://doi.org/10.1016/j.jclepro.2013.01.017
- Vattenfall. (n.d.). Our CO2 roadmap towards climate neutrality. Vattenfall. Retrieved May 21, 2023, from https://group.vattenfall.com/what-we-do/roadmap-to-fossil-freedom/co2roadmap
- Vattenfall. (2019). HYBRIT orders Norwegain electrolyzers for fossil free steel production in Luleå. Vattenfall. https://group.vattenfall.com/press-and-media/pressreleases/2019/hybritorders-norwegain-electrolyzers-for-fossil-free-steel-production-in-lulea
- Vogl, V. (2023). Steel Beyond Coal: Socio-Technical Change and the Emergent Politics of Steel Decarbonisation. Lund University, Environmental and Energy Systems Studies. https://lucris.lub.lu.se/ws/portalfiles/portal/134115357/e_nailing_ex_Vogl.pdf

- Vogl, V., Åhman, M., & Nilsson, L. J. (2018). Assessment of hydrogen direct reduction for fossil-free steelmaking. *Journal of Cleaner Production*, 203, 736–745. https://doi.org/10.1016/j.jclepro.2018.08.279
- Vogl, V., Åhman, M., & Nilsson, L. J. (2021). The making of green steel in the EU: A policy evaluation for the early commercialization phase. *Climate Policy*, 21(1), 78–92. https://doi.org/10.1080/14693062.2020.1803040
- Wood, H. (2023, August 3). What Is Rebar? *Materials Market*. https://materialsmarket.com/articles/what-is-rebar/
- World Steel Association. (2021). Top Producers. *Worldsteel.Org.* https://worldsteel.org/steel-topics/statistics/top-producers/
- World Steel Association. (2022). World Steel in Figures 2022. Worldsteel.Org. https://worldsteel.org/steel-topics/statistics/world-steel-in-figures-2022/
- Yi, S.-H., Choi, M.-E., Kim, D.-H., Ko, C.-K., Park, W.-I., & Kim, S.-Y. (2019). FINEX as an environmentally sustainable ironmaking process. *Ironmaking & Steelmaking*, 46(7), 625–631. https://doi.org/10.1080/03019233.2019.1641682

Appendix

Appendix 1. Benchmark value for sinter (Source: European Commission, 2021)



Key parameters for BM3 Sintered Ore	Value	Unit
Average GHG emissions intensity of the 10% most efficient installations in 2016/2017	0.163	t CO2e/t
Benchmark value for 2021-2025	0.157	t CO2e/t
Benchmark value for phase 3, 2013-2020 (as a reference)	0.171	t CO2e/t
Average GHG emissions intensity of all installations in 2016/2017	0.259	t CO2e/t
Weighted average GHG emissions intensity of all installation in 2016/2017	0.248	t CO2e/t
(Attributed) GHG emissions covered by benchmark (average of 2016/2017)	23 154 801	t CO2e/t
Preliminary free allocation covered by benchmark in 2021	14 393 515	EUA



Appendix 2. Benchmark value for hot metal (Source: European Commission, 2021)

Key parameters for BM4 Hot metal	Value	Unit
Average GHG emissions intensity of the 10% most efficient installations in 2016/2017	1,331	t CO2e/t
Benchmark value for 2021-2025	1,288	t CO2e/t
Benchmark value for phase 3, 2013 – 2020 (as a reference)	1,328	t CO2e/t
Calculated update rate in %/year for the period from 2007/2008 to 2016/2017	0,02%	
Update rate in %/year applied to the phase 3 benchmark	-0,20%	
Update rate in % applied to the phase 3 benchmark	-3,0%	
Median GHG emissions intensity of all installations in 2018/2017	1,443	t CO2e/t
Average GHG emissions intensity of all installations in 2016/2017	1,520	t CO2e/t
Weighted average GHG emissions intensity of all installations in 2016/2017	1,495	t CO2e/t
Number of (sub-)installations using the benchmark for free allocation	28	
Number of (sub-)installations taken into account for the benchmark value update	25	
(Attributed) GHG emissions covered by benchmark (average of 2016/2017)	128 889 587	t CO2e
Preliminary free allocation covered by benchmark in 2021	110 024 564	EUA

Appendix 3. Interview Questions

Interview questions to (organization)

The purpose of my thesis research is to investigate the internal and external environments of Sweden and Korea with respect to the transition to decarbonized steel technology. It aims to identify the opportunities and challenges involved, and to suggest the appropriate direction for both countries to introduce and scale up green steel technology. This interview with (organization) will be greatly invaluable and helpful to learn practitioners' perspective & insights into the practical aspects of the situation in Sweden. You don't need to answer them all if it's not relevant to your field. The interview responses will be used as data solely for my thesis. Please feel free to share and add more of your insights and opinions that <u>is</u> not covered by these interview questions. I appreciate your time and participation.⁴⁴

÷

Below will be brief of interview questions, including but not limited to:

- How would you describe (organization) top 2-3 roles in Sweden and in the EU in terms of green steel?
- In the HYBRIT collaboration, academia and other organizations have collaborated in working packages. How was research collaboration successfully established?
- How do you foster human resources in terms of green steel in Sweden?
- How is the political network in terms of green steel in Sweden?
- What is your opinion about the current discussion on the global standard (e.g. Responsible steel) on green steel in Sweden?⁽⁴⁾
- In your opinion, what has been the most effective political support (<u>e.q.</u> policy, etc.) implemented in Sweden that accelerates green steel?⁴⁴
- Are there any areas where you think policy improvement should be made to boost steel industry decarbonization in Sweden?
- In your opinion, what kind of tasks/roles should be prioritized from the Swedish government / steel producers / (organization) / other relevant stakeholders to boost the green steel transition in Sweden & EU?44
- What are the main challenges in terms of boosting green steel transition in Sweden?
- 10. Is there any support scheme to create a market / demand for green steel in Sweden?4